Düsseldorf Institute for Competition Economics

DISCUSSION PAPER

No 127

The Rise of the East and the Far East: German Labor Markets and Trade Integration

Wolfgang Dauth, Sebastian Findeisen, Jens Suedekum

December 2013



d|u|p düsseldorf university press

IMPRINT

DICE DISCUSSION PAPER

Published by

düsseldorf university press (dup) on behalf of Heinrich-Heine-Universität Düsseldorf, Faculty of Economics, Düsseldorf Institute for Competition Economics (DICE), Universitätsstraße 1, 40225 Düsseldorf, Germany www.dice.hhu.de

Editor:

Prof. Dr. Hans-Theo Normann Düsseldorf Institute for Competition Economics (DICE) Phone: +49(0) 211-81-15125, e-mail: <u>normann@dice.hhu.de</u>

DICE DISCUSSION PAPER

All rights reserved. Düsseldorf, Germany, 2013

ISSN 2190-9938 (online) - ISBN 978-3-86304-126-7

The working papers published in the Series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editor.

The Rise of the East and the Far East: German Labor Markets and Trade Integration *

Wolfgang DauthSebastian FindeisenIABUniversity of Mannheim

Jens Suedekum University of Duisburg-Essen

December 2013

Abstract

We analyze the effects of the unprecedented rise in trade between Germany and "the East" – China and Eastern Europe – in the period 1988–2008 on German local labor markets. Using detailed administrative data, we exploit the cross-regional variation in initial industry structures and use trade flows of other high-income countries as instruments for regional import and export exposure. We find that the rise of "the East" in the world economy caused substantial job losses in German regions specialized in import-competing industries, both in manufacturing and beyond. Regions specialized in export-oriented industries, however, experienced even stronger employment gains and lower unemployment. In the aggregate, we estimate that this trade integration has caused some 442,000 additional jobs in the economy and contributed to retaining the manufacturing sector in Germany. This is almost exclusively driven by the rise of Eastern Europe, not by China. We also conduct an analysis at the individual worker level, and find that trade had a stabilizing overall effect on employment relationships.

JEL-Classification: F16, J31, R11 Keywords: International Trade, Import Competition, Export Opportunities, Local Labor Markets, Employment, China, Eastern Europe, Germany

^{*}Email contacts: wolfgang.dauth@iab.de, sebastian.findeisen@uzh.ch, jens.suedekum@uni-due.de. We thank Stefano DellaVigna, David Autor, Marius Brülhart, Bernd Fitzenberger, William Kerr, Daniel Sturm, Fabrizio Zilibotti, and participants of various seminars for very helpful comments. The usual disclaimer applies.

1 Introduction

One of the central forces of globalization in the last decades has been the rise of Eastern Asian countries, especially China, in the world economy. The substantial rise of trade with China, and its perceived competitiveness, have led to major concerns in Western market economies about possible adverse effects for domestic labor markets. This "fear" is particularly strong in the United States, and numerous studies have addressed the impacts of this trade integration on the US economy.¹



Figure 1: German trade volumes with China and Eastern Europe, 1978-2008.

From the perspective of Germany, which consistently ranks among the most open economies in the world and for a long time held the unofficial title of the "export world champion", China's rise was also striking. Starting from almost zero trade until the late 1980s, the German annual import volume from China has risen to more than 50 billion Euros in 2008 (see Figure 1). This corresponds to a growth rate of 1608 percent, which is far higher than for any other trading partner. However, although Germany runs a trade deficit vis-a-vis China despite an overall trade surplus, the magnitude of this deficit is much smaller than in the US case. This is because German exports to China have also risen by about 900 percent, from almost zero in 1988 to some 30 billion Euros in 2008. The "rise of China" has therefore led to two major changes for the German economy: Increased import competition, particularly in such sectors as textiles, toys, or office and computer equipment, but at the same time a substantial rise in market opportunities for German exporters, most notably in sectors such as automobiles, specialized machineries, and electronic and medical equipment.

In addition to the "rise of China", Germany was affected by another major facet of globalization that – at least economically – had a much milder impact in North America, namely the sudden and unexpected fall of the iron curtain with the subsequent transformation of the former socialist countries into market economies. Overall, the rise of German exports to Eastern Europe even outpaced export growth to China. Im-

¹See, among others, Feenstra and Hanson (1999); Harrigan (2000); Feenstra and Wei (2010); Harrison et al. (2010); Ebenstein et al. (2013).

port growth also has been substantial, exceeding 800 percent during the period 1988-2008.² For the German economy, trade exposure therefore increased not only from the Far East, but also from the East closer by.

In this paper, we analyze the impacts of these major trade shocks from the perspective of small-scale German regions. There is substantial variation in sectoral employment patterns at the regional level, also within the manufacturing sector where commodity trade occurs. Given these initial specializations, regions are differently exposed to import competition and export opportunities arising from Eastern European and Asian countries. Regions that are strongly specialized in export-oriented industries, say "automobile regions", may benefit from the rise of new markets, while regions specialized in import industries, say "textile regions", may see their labor markets put under strain by the rising exposure to foreign competition. In our aggregate analysis, we relate changes in key local labor market variables to measures of import and export exposure that reflect the local industry mix. Afterwards, we adopt a disaggregate approach and analyze how trade exposure affects individual employment biographies across heterogeneous worker-establishment pairs in different industries.

This study is most closely related to a recent string of literature that identifies the impact of trade shocks at the *regional level*, see Chiquiar (2008), Kovak (2011), Topalova (2010) and in particular, Autor, Dorn and Hanson (2013) (henceforth labeled as *ADH*). They separate the US into local labor market areas and analyze the differential performance of these regions depending on their initial industry employment patterns, which determine their exposure to import competition from China. To account for unobserved shocks that simultaneously affect imports and regional performance, they use trade flows of other high-income countries as an instrument for US local trade exposure. Their main finding is that regions strongly prone to Chinese import competition have experienced severe negative impacts on their labor markets, such as lower manufacturing employment, rising unemployment, or lower labor force participation.³

Our empirical approach is similar to *ADH*, but we obtain results for Germany that differ substantially from their findings. Consistent with the US experience, we also find a negative *causal* effect of import exposure from the East. That is, German regions specialized in import competing sectors saw a decline in manufacturing employment attributable to the impact of trade. Yet, this negative impact is, on average, more than offset by a positive *causal* effect of export exposure, as the respective export oriented regions built up manufacturing employment as a result of the new market opportuni-

²We consider Eastern Europe to comprise the countries Bulgaria, Czech Republic, Hungary, Poland, Romania, Slovakia, Slovenia, and the former USSR or its succession states Russian Federation, Belarus, Estonia, Latvia, Lithuania, Moldova, Ukraine, Azerbaijan, Georgia, Kazakhstan, Kyrgyzstan, Tajikistan, Turkmenistan, and Uzbekistan. The increase in trade volumes between the US and these countries is negligible, at least in comparison to the German numbers.

³*ADH* also find that Chinese trade shocks induced only small cross-regional population shifts. This low labor mobility, in turn, supports the view that regions can be treated as "sub-economies" across which adjustment to shocks works far from instantaneously. Since regional labor mobility in Germany is traditionally much lower than in the US (Molloy et al.; 2011), (Bertola; 2000), their empirical approach indeed seems especially well applicable in the German context.

ties. In the aggregate, we estimate that the rise of the East has retained some 442,000 full-time equivalent jobs in Germany over the period 1988–2008 that would not exist without this trade integration. This aggregate implication is very different from *ADH*'s conclusions for the US, and leads to a starkly opposite picture how globalization has affected the labor markets in Western market economies.

We also pay close attention to the question which part of "the East" – China or Eastern Europe – had stronger impacts in German local labor markets. The rise of China, facilitated by several market openings and reforms, productivity gains of Chinese producers, and market access gains after its WTO accession in 2001, occurred quickly and exogenously from the point of view of "the West". Yet, the rise of Eastern Europe shared many of those characteristics. The fall of the iron curtain in the late 1980s was arguably at least as sudden as the Chinese market openings, and the transformation of the former socialist block triggered substantial productivity gains in those economies (Burda and Severgnini (2009)). Moreover, many Eastern European countries adopted concrete steps of trade integration early on, for example in 1995, where several of them (including Poland, Czech Republic and Hungary) joined the WTO.⁴ The increasing German export and import volumes documented in Figure 1 are, thus, likely to stem – at least to a large extent – from the strongly rising productivity and accessibility of Eastern Europe, particularly during the first decade after the iron curtain fell, and less so by developments that have their origin in Germany. And indeed, several studies have exploited this unique historical setting as a quasi-natural experiment, including Redding and Sturm (2008), Brülhart et al. (2012), or Glitz (2012), who investigate how locations in "the West" were affected along different margins by this major exogenous shock.

Our empirical findings suggest that the rise of Eastern Europe affected local labor markets in Germany more strongly than the rise of China. This differential impact seems to be mainly driven by the structure of imports from the two areas. More specifically, we find that changes in trade exposure in such industries where imports from China grew the most (textiles, for example) had negligible labor market effects. The reason seems to be that Germany already tended to import those labor-intensive goods in the 1980s, in which China subsequently became the world's dominant supplier. When the Chinese rise gained momentum, this has then mainly led to a diversion of German import flows from other countries (such as Italy or Greece), but it has not caused major job displacements in Germany. For the case of Eastern Europe, we find a rather different pattern: Germany initially tended to export goods where the subsequent Eastern European rise was particularly strong. Hence, we find substantial displacement effects from rising Eastern European import penetration across German regions. Yet, in the aggregate, those employment losses were more than offset by the creation of new manufacturing jobs stemming from rising German exports to that area. This is quite in contrast to the US experience, where rising import penetration from China fuelled a large overall trade deficit and hurt domestic workers on balance.

⁴In the online appendix we provide additional historical background on the trade integration of Eastern European countries since the 1990s, and the rise of productivity there.

Moving beyond the manufacturing sector, we then investigate how local labor markets responded more broadly to the rise of the East. We find that export-oriented regions saw significant total employment gains and reductions in unemployment. The positive trade effects also spill over to non-manufacturing sectors, such as business services, and cause mildly positive wage gains. Import-competing regions, on the other hand, not only lost manufacturing jobs but were adversely affected more generally.

Finally, our worker-level analysis allows for an even more detailed look at the causal effects of trade. Here, we use cumulative employment spell information and find that a higher import exposure lowers expected overall employment durations, and also the workers' expected job tenure with the original employer or within the initial local industry. Overall, however, we find that rising trade exposure has *stabilized* individual employment biographies, since rising export exposure tends to raise expected employment durations. This pattern is quite stable across different types of workers (by age, gender and qualification), and we find that trade shocks seem to affect employees of small establishments more strongly. We then discuss those findings in the light of recent theories of international trade with heterogeneous firms.

The rest of this paper is organized as follows. Section 2 describes the empirical approach and the data. Sections 3 and 4 are devoted to the analysis of manufacturing employment at the regional level, while Section 5 looks at other local labor market outcomes. Section 6 presents the worker-level analysis, and Section 7 concludes.

2 Empirical approach

2.1 Trade exposure across local labor markets

Our empirical strategy exploits the variation in initial industry specialization across local labor markets at the onset of the economic rise of "the East", i.e., China and Eastern Europe. We first consider the import exposure of a German region *i*. Using *ADH*'s approach, which is based on a monopolistic competition model of international trade with cross-country productivity differences, this import exposure can be written as:

$$\Delta (Import \ exp.)_{it}^{EAST} = \sum_{j} \frac{E_{ijt}}{E_{jt}} \frac{\Delta Im_{jt}^{D \leftarrow EAST}}{E_{it}}, \tag{1}$$

where $\Delta I m_{jt}^{D \leftarrow EAST}$ is the total change in imports from "the East" to Germany that was observed in industry *j* between time periods *t* and t + 1.5 E_{ijt}/E_{jt} represents region *i*'s share of national industry employment in *j* at time *t*, and E_{it} is total manufacturing employment in region *i*. This measure in (1) thus captures the *potential* increase in import exposure of a German region *i*, given its initial sectoral employment structure, as it apportions the *national* change in sectoral imports to the single regions according to their shares in national industry employment.

⁵For now, $\Delta Im_{jt}^{D \leftarrow EAST}$ refers to the joint increase of German imports from both China and Eastern Europe. In Section 4 we also consider import exposure from the two areas separately.

To capture regional export exposure, we derive an analogous measure:

$$\Delta(Export \ exp.)_{it}^{EAST} = \sum_{j} \frac{E_{ijt}}{E_{jt}} \frac{\Delta E x_{jt}^{D \to EAST}}{E_{it}},\tag{2}$$

which captures the potential of a region, given its initial sectoral employment patterns, to benefit from rising demand from the East for German manufacturing products. In the empirical analysis we aim to identify the causal effect of the rise of the East on the economic performance of German regions. More specifically, we regress the change of a regional outcome variable (such as overall manufacturing employment) between t and t + 1, denoted as ΔY_{it} , on the change of regional import and export exposure over the same time period, see eqs. (1) and (2), while controlling for start-of-period regional control variables X'_{it} as discussed below:

$$\Delta Y_{it} = \beta_0 + \beta_1 \cdot \Delta (Import \ exp.)_{it}^{D \leftarrow EAST} + \beta_2 \cdot \Delta (Export \ exp.)_{it}^{D \rightarrow EAST} + X'_{it} \cdot \beta_3 + \varepsilon_{it}.$$
(3)

The main challenge for this exercise is the endogeneity of trade exposure, in particular the presence of unobserved supply and demand shocks that simultaneously affect import/export exposure and regional economic performance.

2.2 Identification strategy

To address this concern, we use an instrumental variable (IV) strategy that is close in spirit to the approach by *ADH*. To instrument German regional import exposure from the East, we construct the following variable for every German region *i*:

$$\Delta(Import \, exp._{Inst})_{it}^{EAST} = \sum_{j} \frac{E_{ijt-1}}{E_{jt-1}} \frac{\Delta Im_{jt}^{Other \leftarrow EAST}}{E_{it-1}}.$$
(4)

Here, $\Delta Im_{jt}^{Other \leftarrow EAST}$ are changes in import flows of industry *j*-goods from the East to other countries. Similarly, for regional export exposure we construct the following instrumental variable that uses changes in exports of other countries to the East:

$$\Delta(Export \ exp._{Inst})_{it}^{EAST} = \sum_{j} \frac{E_{ijt-1}}{E_{jt-1}} \frac{\Delta E x_{jt}^{Other \to EAST}}{E_{it-1}}.$$
(5)

The idea behind the instrument (4) is that the rise of China and Eastern Europe in the world economy induces a supply shock and rising import penetration for *all* trading partners, not just for Germany. Using the import flows of other countries as an instrument for local import exposure in Germany therefore identifies the exogenous component of rising competitiveness in the East, and purges the effects of possible shocks that simultaneously affect German imports and regional performance variables.⁶

⁶Note that the import values are distributed across German regions according to *lagged* sectoral employment shares. This is done to tackle issues of measurement error or reverse causality, if employment reacted in anticipation of future trade exposure.

The logic of the instrumental variable (5) is similar. As the East rises in the world economy, this induces a demand shock for *all* countries since the East becomes a more attractive export destination not just for Germany. Using (5) as an instrument for (2) purges the impacts of unobservable shocks, and thus identifies the causal impact of the rise of export opportunities in the East on German local labor markets.

The quality of the instruments hinges, in particular, on three important conditions. First, they should have explanatory power in order to avoid a weak instrument problem. Second, the unobservable supply and demand shocks in those other countries should not be strongly correlated with those of Germany, since otherwise the instruments do not purge those shocks and the estimated coefficients would still be biased. Third, in order for the exclusion restriction not to be violated, there should not be independent effects of the trade flows of those other countries with China/Eastern Europe on the German regions, other than through the exogenous rise of the East.

To take these conditions into account, it is important to consider which countries are included in the "instrument group" whose trade flows are used to construct (4) and (5). We adopt the following approach: We focus on developed countries with a similar income level as Germany, but we exclude all direct neighbors as well as all members of the European Monetary Union. This is for two reasons. First, supply and demand shocks in such countries (e.g., France or Austria) are likely to be too similar to those in Germany, hampering the identification. Second, since those countries are highly integrated with Germany in an economic union where exchange rate alignments are impossible, it is likely that shocks which change trade flows between those countries and the East also directly affect regional performance in Germany, thus violating the exclusion restriction. For this reason we also do not consider the United States in the benchmark group, because of its high significance in the world economy.

Our final instrument group consists of Australia, Canada, Japan, Norway, New Zealand, Sweden, Singapore, and the United Kingdom. The trade flows of those countries with the East are, indeed, relevant instruments for German trade exposure according to our first-stage results reported below. Moreover, we expect correlations in demand and supply shocks between Germany and those countries, and possible independent effects of shocks in those countries on German local labor markets, to be relatively modest. Still, we conduct several robustness checks where we change the instrument group. Furthermore, we also consider an alternative identification strategy (described in the online appendix) based on a gravity approach to international trade. More specifically, instead of using *actual* increases in German imports and exports as in (1) and (2), which may also be driven by shocks, we use gravity residuals that capture the *implied* differential rise of competitiveness and market access of the East, and respectively, the *implied* differential rise of attractiveness of the East as a market for German exports, relative to other possible origins/destinations.⁷

⁷This approach thus extends the gravity-based method by *ADH* by constructing a regional exposure measure not only for imports but also for (gross and net) exports.

2.3 Data

For the analysis at the regional level, we combine two main data sources. The German labor market data at the regional and local industry level come from the IAB-Establishment History Panel (BHP, see Spengler; 2008) which includes the universe of all German establishments with at least one employee subject to social security. This data set consists of an annual panel with approximately 2.7 million yearly observations on establishments aggregated from mandatory notifications to social security in the years from 1975 to 2008. Due to the administrative origin, the data are restricted to information relevant for social security (structure of workforce with regard to age, sex, nationality, qualification, occupation, wage) but at the same time are highly reliable and available on a highly disaggregated level.

Detailed data for regional sectoral employment is available from 1978 onwards. Since much of the rise of China and Eastern Europe occurred after 1990, we use 1988 as our starting point and thus observe data for two time periods (1988 to 1998 and 1998 to 2008) for each region. This timing also allows us to use employment lagged by ten years in the construction of our instruments as discussed above. Eastern German regions are only included for the second decade 1998 to 2008, because sectoral employment data for these regions only became available in the mid-1990s.

Information on international trade is taken from the United Nations Commodity Trade Statistics Database (Comtrade). This data contains annual international trade statistics of over 170 reporter countries detailed by commodities and partner countries. Trade flows are converted into Euros of 2005 using exchange rates supplied by the German *Bundesbank*. We merge these two data sources by harmonizing industry and product classifications. The correspondence between 1031 SITC rev. 2/3 product codes and the employment data (equivalent to NACE 3-digit industry codes) is provided by the UN Statistics Division and allows unambiguously matching 92 percent of all commodities to industries. Trade values of ambiguous cases are partitioned into industries according to national employment shares in 1978. Finally, we drop all industries related to agriculture, mining and fuel products, so that our empirical analysis focuses on 101 different manufacturing industries.

2.4 Descriptive overview

In Appendix Tables (A.1) and (A.2) we report the sectors with the highest import and export volumes vis-a-vis Eastern Europe and China, respectively. As can be seen, Germany tends to export similar goods to the two areas, most notably cars, car parts, and special purpose machinery. Yet, the imports are rather different: From China, Germany mainly imports textiles, toys, and lower-tier electronic products. In the case of Eastern Europe, on the other hand, the top import sectors are also motor vehicles and car parts, that is, there is substantially more intra-industry trade with this area.

The changes of sectoral trade volumes over time are used to construct the regional exposure measures according to (1) and (2), and Appendix Table (A.3) reports the mean

values (and standard deviations) across all German regions separately for the two time periods and the two areas. For China, we see that German regions were exposed to trade mainly during the second period (1998-2008). On average, the exposure to Chinese imports rose by $\leq 1,900$ during those ten years, as compared to only ≤ 590 between 1988 and 1998. Export exposure also increased much faster in the second time period.

For Eastern Europe, the average import and export exposure of German regions increased even stronger and also substantially earlier. Specifically, the import exposure rose by \in 1,800 and the export exposure by \in 2,170 between 1988 and 1998. Then, during the second period, trade volumes continued to grow and average export exposure (\in 3,710) outpaced import exposure (\in 1,850) even more clearly. Notice an interesting parallel in the rise of the two areas. As is apparent from these numbers as well as from Figure 1 above, trade flows with Eastern Europe particularly accelerated since the mid-1990s. Around that time, many important Eastern European countries joined the WTO (also see the online appendix), and this picture is thus consistent with the substantial expediting of trade with China in the aftermath of its WTO accession in 2001.

Behind these average exposures, there is substantial variation across local labor markets in Germany as illustrated in Figures A.1 and A.2 in the online appendix.⁸ It stands out that former Eastern Germany faced little import competition or export opportunities. The reason is the massive de-industrialization that occurred quickly after reunification, due to the strong real appreciation that came with the introduction of the *Deutschmark*. Western Germany was exposed much more strongly to trade, as the manufacturing sector is concentrated there, and within the West, exposure varies considerably across regions depending on the detailed local employment patterns.

Finally, Appendix Table (A.4) reports the cross-regional correlations of local import and export exposure vis-a-vis China and Eastern Europe, respectively. Bearing in mind that aggregate German exports to the two areas consist of similar goods, it is not surprising to see a high correlation (0.835) between the two regional measures of export exposure. For import exposure, this correlation is much weaker (0.328), indicating that some German regions have industrial structures that make them vulnerable to import competition from one area, but not the other. Moreover, bearing in mind the strong prevalence of intra-industry trade with Eastern Europe helps to understand the high correlation (0.745) between local import and export exposure with respect to this area. For China, this correlation is much lower (0.394) since German exports to, and imports from China consist of very different types of goods.

3 Trade exposure and manufacturing employment

We now turn to our econometric analysis, where we estimate eq. (3) and use (4) and (5) as instruments for our main variables (1) and (2). Because of the partly substantial correlations just described it would not be sensible, however, to control at the same

⁸The maps focus on the period 1998-2008, so that former Eastern Germany is also included.

time for import and export exposure with respect to both China and Eastern Europe. We rather adopt a complimentary approach. Specifically, we first add up China and Eastern Europe and analyze how German local labor markets were affected by the joint rise of "the East", distinguishing the impacts from import and export exposure. Afterwards, we consider the effects of *net* exposure and distinguish the effects from the two areas, China and Eastern Europe.⁹

Our benchmark dependent variable is the decennial change in manufacturing employment as a share of the working age population in region i, $Y_{it} = E_{it}^{M/WP}$. In Section 5, we then consider other local labor market indicators as outcome variables.

3.1 Baseline Specification: Manufacturing Employment Growth

The first column of Table (1) shows the most parsimonious specification, where the only additional control is the overall regional employment share in tradeable goods industries.¹⁰ This variable is important, since our approach seeks to exploit the detailed variations of employment structures *within* the manufacturing sector. The first-stage results reported in the bottom panel indicate that our instruments are strong, with F-test statistics well above conventional threshold levels. Turning to the second-stage results, export exposure has a positive and significant impact on manufacturing employment growth, while the impact of import exposure is not statistically different from zero. We also find a trend of mean reversion of manufacturing employment, since growth is negatively related to the initial share of tradeable goods industries.

In column 2 we add control variables for the composition of the local workforces: A higher initial share of high-skilled, foreign and female workers is negatively related to manufacturing employment growth, since those groups are more prevalent in service industries. Furthermore, motivated by the literature on job off-shoring (e.g. Grossman and Rossi-Hansberg; 2008), we include the percentage of routine-intensive occupations (represented by basic activities in the taxonomy of Blossfeld (1987)), but find no significant impact. Most importantly, the inclusion of these controls leaves the results for our central variables unaffected.

In column 3 we add dummy variables for the time periods and for four higherorder German regions (North, South, West, East) comparable to Census divisions in the US. The coefficients for export exposure as well as for the other control variables remain stable. Moreover, the negative impact of import exposure on manufacturing employment gets larger in absolute terms and is now also statistically significant.

Next, to address the special role of cars for the German economy, we split up the initial share of tradable goods industries into two parts: the initial local employment share in the automotive industry, and the initial share of all other tradable goods sectors. In column 4, we find that there is mean reversion in both manufacturing branches.

⁹The correlation between net expost exposure to China and Eastern Europe across all German regions is virtually zero (-0.067), thus allowing for a separate identification of the effects.

¹⁰To account for spatial and serial correlation, we cluster the standard errors at the level of 50 greater labor market areas as defined in Kropp and Schwengler (2011) in all specifications.

	Dependent variable: 10-year change manufacturing employment / working age pop. in %-points						
	(1)	(2)	(3)	(4)	(5)	(6)	
	(2SLS)	(2SLS)	(2SLS)	(2SLS)	(2SLS)	(OLS)	
Δ import exposure	0.035	-0.028	-0.149**	-0.158**	-0.190***	-0.075	
	(0.09)	(0.08)	(0.07)	(0.07)	(0.07)	(0.05)	
Δ export exposure	0.332***	0.444***	0.409**	0.425**	0.399*	0.442***	
	(0.12)	(0.12)	(0.18)	(0.21)	(0.21)	(0.12)	
% manuf. of tradable goods	-0.108***	-0.083***	-0.073***				
	(0.01)	(0.02)	(0.02)				
% high skilled		0.011	-0.046	-0.042	-0.039	-0.047	
		(0.05)	(0.04)	(0.04)	(0.04)	(0.04)	
% foreigners		-0.197***	-0.159***	-0.159***	-0.162***	-0.161***	
		(0.04)	(0.04)	(0.04)	(0.04)	(0.03)	
% women		-0.057***	-0.062***	-0.061***	-0.060***	-0.060***	
		(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	
% routine occupations		-0.039	-0.025	-0.021	-0.014	-0.023	
		(0.03)	(0.03)	(0.03)	(0.03)	(0.03)	
% manuf. of other tradable goods				-0.073***	-0.068***	-0.083***	
				(0.02)	(0.02)	(0.02)	
% manuf. of cars				-0.081**	-0.074**	-0.092***	
				(0.03)	(0.03)	(0.03)	
Time dummy	_	_	Yes	Yes	_	_	
Region dummies	_	_	Yes	Yes	_	_	
Region \times time interaction	-	_	_	-	Yes	Yes	
First stage results, dependent varia	ble: Δ imp	ort exposur	e				
Δ import exposure	0.255***	0.257***	0.248***	0.253***	0.252***		
(other countries)	(0.05)	(0.05)	(0.06)	(0.06)	(0.06)		
Δ export exposure	0.146***	0.144***	0.117***	0.107***	0.105***		
(other countries)	(0.03)	(0.03)	(0.03)	(0.04)	(0.03)		
R-squared	0.751	0.756	0.761	0.762	0.763		
F-test of excluded instruments	105.809	74.453	17.984	18.127	16.361		
First stage results, dependent varia	ble: $\Delta \exp(\frac{1}{2})$	ort exposure	9				
Δ import exposure	0.038*	0.035*	-0.006	0.006	0.002		
(other countries)	(0.02)	(0.02)	(0.01)	(0.01)	(0.01)		
Δ export exposure	0.567***	0.555***	0.423***	0.398***	0.393***		
(other countries)	(0.05)	(0.05)	(0.06)	(0.06)	(0.06)		
R-squared	0.766	0.772	0.818	0.824	0.826		
F-test of excluded instruments	96.909	81.708	26.050	25.116	22.280		

Table 1: Trade Exposure and Manufacturing Employment

Notes: N=739 (326 regions in first and 413 regions in second period). All regressions include a constant. Both first stage regressions include the same set of control variables as the corresponding second stage in the upper panel. All control variables are constructed as the number of workers in a particular group relative to total employment. % manuf. of tradable goods is the percentage of workers in manufacturing of consumption, production, or capital goods. % high skilled is the percentage of workers with a university degree. % routine occupations is the percentage of workers in basic unskilled manual, service, and administrative occupations according to the taxonomy of Blossfeld (1987). Standard errors clustered at the level of 50 aggregate labor market regions in parentheses. Levels of significance: *** 1 %, ** 5 %, * 10 %.

Moreover, our key results for export and import exposure remain robust.

Finally, in column 5 we use interacted region \times time period fixed effects instead of separate dummies. This specification is the most demanding one, as it is only identified by within region-time variation. Again, we find that our results remain stable. Moreover, the first-stage results remain highly significant throughout.

For this benchmark specification, we also reports the results of a simple ordinary least squares (OLS) regression (see column 6). The comparison with the two-stage least squares (2SLS) coefficient from column 5 shows that the OLS estimate for import exposure is biased towards zero due to measurement error and unobserved demand shocks.¹¹ For export exposure there are two countervailing effects: a downward bias due to measurement error, and an upward bias due to the impact of unobserved supply shocks. The second source of bias seems to be somewhat more important, but the OLS estimate is roughly in the same ballpark as the baseline 2SLS coefficient.

Robustness checks: Regional classification. We have conducted a battery of sensitivity tests that deal with the regional units used in our analysis. First, we drop all regions located in former Eastern Germany and repeat our analysis only for the local labor markets in the Western part. This is an important robustness check, since the two parts of Germany have been very different in terms of industry structure and economic performance, and our time period coincides with many other important changes. The results are reported in columns 1 and 2 of the upper panel in Appendix Table A.5. Reassuringly, both for import and export exposure, the effects become even stronger when focussing only on Western German regions.

Another concern might be that the 413 counties (Landkreise and kreisfreie Städte) may be too small to constitute independent local labor markets. In columns 3 and 4 of the upper panel in Appendix Table A.5, we thus carry out our analysis at the level of 147 functional labor markets (delineated according to commuting distances, see Eckey et al. (2006)). Re-estimating our benchmark specification, we obtain qualitatively unchanged results although the coefficient for export exposure is reduced. Finally, we use dummies for the 16 German Federal States instead of the higher-order regions, and respectively, State \times time interactions. The results are reported in Appendix Table A.6, and turn out to be very similar to our baseline findings.

Robustness checks: Instrument group. In the lower panel of Appendix Table A.5, we check the sensitivity with respect to the construction of our instruments. First, we specify an over-identified model instead of the just identified 2SLS model used as the benchmark. This approach does not add up the import/export flows of all instrument countries with the East, but exploits the detailed variation across countries. The results remain qualitatively unchanged, and the Hansen J test statistic of 20.276 (p=0.12) does not reject the hypothesis of validity of the excluded instruments.

¹¹For the US case, *ADH* also find that the OLS coefficient is smaller in absolute terms than the 2SLS coefficient, so our results are consistent with this. In the next subsection we provide a detailed comparison of our results and theirs.

Next, we change the countries that are included in the instrument group, in order to address the concern that there could be independent direct effects of shocks in those countries on German regions. This may be particularly relevant for the United Kingdom, which is the most important trading partner of Germany among the benchmark instrument countries. We therefore drop the UK from the group and re-estimate the (just identified) baseline specification. The estimation results are almost the same as before. We next add the USA to the instrument group, but again this hardly affects the results. Finally, we add several neighboring countries (such as France). The explanatory power of the instrument gets even stronger in that case, particularly for export exposure, and second-stage results remain robust. But for this specification one surely has to worry about the exclusion restriction.

Overall, the lower panel of Appendix Table A.5 suggests that our results are robust with respect to specification issues of the instrumental variable approach.

3.2 Quantitative comparison to the US and further robustness checks

What do the benchmark results imply quantitatively? Our preferred estimates from column 5 in Table (1) imply that a 10-year change of $\leq 1,000$ per worker in import exposure reduces manufacturing employment relative to working age population by 0.19 percentage points, whereas export exposure increases this share by 0.4 percentage points. Taking into account that average export exposure per worker increased by $\leq 7,060$ from 1988-2008, and import exposure by $\leq 6,147$, we can calculate that the new export opportunities in the East increased the aggregate manufacturing share by 2.82 percentage points. Import competition reduced it by only 1.17 percentage points, thus leading to a net increase by 1.65 percentage points as a result of trade integration.

To set this number into perspective, note that the overall manufacturing employment share in the working-age population has been declining in Germany over the period 1988-2008 by about 4 percentage points, from roughly 16 % to 12 %, reflecting a general trend of structural change observed in most developed countries. Our results therefore indicate that trade integration with Eastern Europe and China has slowed down this trend, and has retained manufacturing in the German economy.

Comparison to the US. The results by *ADH* lead to a rather different picture. They estimate a coefficient of -0.596 for the causal effect of Chinese import penetration on manufacturing employment/working-age population in the US (see their Table 3, column 6). Taking into account that their analysis is conducted in 2007-\$ instead of 2005-€, this coefficient can be converted to around -0.786 and then compared to our -0.19 reported in column (5) of Table (1).¹² The marginal impact of import exposure thus seems to be significantly larger in the US than in Germany. Benchmarking the impact of import exposure, *ADH* report that the rise of China has led to a decrease in the

¹²For the conversion we have taken into account an average rise in German prices by +3.9% between 2005 and 2007, and an average exchange rate of 1.3705 \$ per \in in 2007.

manufacturing employment share by 0.68 percentage points in the period 1990-2000, and by 1.10 percentage points in the period 2000-2007. In other words, import exposure from China seems to have caused much larger aggregate job losses in the US than in Germany, where the impact of imports (even with respect to China *and* Eastern Europe together) reduced the manufacturing share by only 1.17 percentage points over 20 years. Moreover, in the German case, those job losses were more than compensated by aggregate job gains stemming from rising export opportunities in the East, something that apparently did not happen in the US.

As noted by *ADH*, those quantitative predictions likely overstate the impact of trade on manufacturing employment. Even though the 2SLS coefficients identify the causal effects of the rise of the East, the measured changes in trade flows may also partly stem from other sources such as unobserved shocks.¹³ However, even if we correct for this and provide a more conservative benchmarking, the principal conclusions are the same: trade with the East has raised the manufacturing share in Germany by 1.222 – 0.627 = 0.595 percentage points, which corresponds to roughly 305,000 manufacturing jobs that were retained. In the US, by contrast, *ADH* estimate that trade exposure with China has destroyed more than 1.5 million manufacturing jobs.

		Dependent variable: 10-year change manufacturing employment / working age pop. in %-points						
	(1) (2) (3) (4) (5) (6)				(7)	(8)		
	(2SLS) Bench	(OLS) 1mark	(2SLS) + third markets	(OLS) gravity	(2SLS)	(OLS) Falsifi	(2SLS) cation	(OLS)
Δ Net exposure	0.237** (0.09)	0.224*** (0.07)	0.205*** (0.08)	0.067*** (0.02)	0.093 (0.06)	0.078 (0.05)	0.043 (0.05)	0.045 (0.04)
First stage results, depe	ndent vari	able: Δ Ne	et exposure					
Δ Net exposure	Net exposure 0.255***				0.185***		0.188***	
(other countries)	(0.05)		(0.05)		(0.03)		(0.04)	
R-squared (first-stage)	0.542		0.499		0.569		0.559	
F-test of excl. inst.	25.886		29.216		29.269		26.759	

Table 2: Be	enchmarking,	robustness and	comparison to	o ADH
	U,			

Notes: N=739 (columns 1–4), N=326 (columns 5+6) and N=323 (columns 7+8). Δ *net exposure* is constructed by subtracting (1) from (2). In column 3, Δ *net exposure* includes import competition from the East in other Eurozone countries weighted by the respective country's share in Germany's total exports (see the online appendix). In column 4, Δ *net exposure* is constructed according to the gravity approach explained in the online appendix. In columns 5–8 the dependent variable is the growth rate of

manufacturing employment over working age population in the period 1978-1988, and Δ *net exposure* refers to 1988–2008, with control variables from 1978. Columns 5 and 6 include all Western German regions. In columns 7 and 8 we drop Groß-Gerau, Fürth and Ulm. All regressions include the full set of control variables from table 1, column 5. Standard errors clustered at the level of 50 aggregate labor market regions in parentheses. Levels of significance: *** 1 %, ** 5 %, * 10 %.

¹³*ADH* develop a method to decompose the total change in trade over time into two parts: i) the part caused by the exogenous rise of the East, and ii) a residual part that may be explained by other causes. This decomposition essentially relies on the comparison of the 2SLS and the OLS coefficients in the baseline specification (see the theory appendix of *ADH* for details). Using our results from columns 5 and 6 of Table (1), this method implies that 43.3 % of the increase in exports is due to the rise of the East, and 53.8 % of the rise in imports. This compares to the US figure of 48% for Chinese import exposure reported by *ADH*. The share of trade that can be attributed to the exogenous rise of the East thus appears to be roughly in the same range in both cases.

Net exports, exposure on third markets, and gravity. To facilitate a further comparison, we now consider *net* exports. In column 1 of Table 2 we report the 2SLS coefficient for German net export exposure vis-a-vis the East, instrumented accordingly with the combined net exports of the other countries to China and Eastern Europe. First-stage results again suggest that we are working with a strong instrument, and the main coefficient of 0.237 is precisely estimated. This should be compared to the coefficient of 0.594 for the net impact of China on the US (see Table 10, panel D in *ADH*).¹⁴ The marginal effect of trade exposure thus again seems to be stronger in the US. Moreover, net exports to the East have increased in Germany, while US net exports to China have decreased over time, thus leading to opposite quantitative predictions how trade has affected manufacturing employment in the two economies.¹⁵

We then take into account that Germany is exposed to rising import penetration from the East not only in its domestic market, but also in other markets in foreign countries. In the online appendix we derive a more encompassing measure of local net export exposure across German regions that includes such displacement effects in third markets. Column 3 shows that the marginal effect of net export exposure on manufacturing employment becomes a bit smaller, which is mainly due to the scaling of the alternative exposure measure, but remains highly significant.

Next, in column 4 we show the results of the alternative identification strategy where we use gravity residuals instead of raw import/export flows to construct the measures of local net export exposure (see the online appendix for details). Notice that we use OLS for this estimation, as the gravity approach per construction sterilizes the confounding effects of possible unobservable shocks. We find that the marginal effect remains highly significant, but its size decreases even further. This is because the rise in gravity residuals captures only the exogenous differential rise in market size and accessibility of the East, which is smaller than the raw increase of German-East trade flows. But once we convert this marginal effect into economic magnitudes, we find that the gravity exercise – which lends further credence to the causal interpretation of our results – leads to virtually identical quantitative predictions as the instrumental variable approach (using the conservative benchmarking explained before).

Falsification test. Finally, as a further important robustness check, we conduct a falsification exercise. Following *ADH*, we regress past employment changes (1978-1988) on future changes in trade exposure (1988-2008), using both OLS and the 2SLS strategy, and including all control variables from our benchmark model (now measured in 1978). The main concern here is that changes in manufacturing employment and trade exposure may be simultaneously driven by a common long-run trend. For example, employment in some manufacturing industries may have been on a secular decline

¹⁴Note that *ADH* consider US net imports instead of net exports to China, hence the change in sign. We then convert their raw coefficient (0.45) into 2005- \in .

¹⁵In column 2 of Table (2) we also report the OLS coefficient for the impact of net exposure. Doing a similar decomposition exercise as above, we find that 41.4 % of net exports arose because of the causal effect of the rise of the East, which is in line with our previous findings.

even before the rise of the East kicked in, and the decreasing domestic production may then have been substituted by imports from the East. Similarly, industries may have boomed even before the mid-1980s, so that export exposure with the East was rather a symptom than a cause of domestic employment gains in manufacturing.

However, this is actually not the case, as columns 5 and 6 of Table 2 show that lagged employment changes do not predict future regional trade exposure. Moreover, upon graphical inspection, we find that these coefficients appear to be strongly driven by three outlier regions with very idiosyncratic experiences.¹⁶ When dropping those three cases from the analysis, the coefficients for the falsification exercise are cut by about half, see columns 7 and 8. By contrast, dropping the three observations from our benchmark specification, the estimates are basically unchanged. Either way, the falsification test suggests that our results are not driven by pre-trends.

4 Eastern Europe versus China

Until now, we have considered the joint impact of "the East" (China and Eastern Europe) on German local labor markets. Yet, as we have outlined above in Section 2.4, the structure of German trade with these two trading partners has been quite different, in particular on the import side. Hence, we now investigate in this section if the "rise of China" and the "rise of Eastern Europe" had different impacts in Germany.

4.1 Differential effects on German manufacturing employment

In panel A of Table 3, we estimate the effects of import and export exposure separately for the two areas.¹⁷ For Eastern Europe, the estimated marginal effects increase substantially compared to the baseline estimates from Table 1. Moreover, the two coefficients for import and export exposure are of similar magnitude and statistically not distinguishable – a finding that is well in line with our previous observation that trade between Germany and Eastern Europe is mostly intra-industry. In stark contrast, for China the coefficient for import exposure is small and only marginally significant, and the coefficient for export exposure is very imprecisely estimated.

A problem with these specifications is omitted variable bias since we consider trade exposure just with respect to one area while leaving out the (potentially relevant) exposure to the other. We tackle this issue in column 1 of panel B in Table 3, where we control for the *net* export exposure of Germany with respect to China and Eastern Europe in the same regression.¹⁸ For Eastern Europe, we find a coefficient for net exports

¹⁶Each of the three regions (Groß-Gerau, Fürth and Ulm) witnessed a strong decline in manufacturing in the 1980s, in each case driven by the decline of one major employer. Groß-Gerau, for example, was the founding place of *Adam Opel AG*, a formerly large German automobile firm that has been taken over by General Motors in 1920. This large plant had to reduce employment sharply during the 1980s, when General Motors imposed its subsidiary not to sell cars outside of Europe any more.

¹⁷The full results including first-stages and controls are reported in Appendix Tables (A.7) and (A.8). ¹⁸Recall that the two net exposure measures are virtually uncorrelated. For consistency, we use the

	Dependent variable: 10-year change manufacturing employment / working age pop. in %-points					
	A: Import and export exposure separately					
	(1)		(2)			
	Eastern Euro	ope	China			
Δ import exposure	-0.973**		-0.149*			
Δ export exposure	(0.34) 0.900*** (0.32)		(0.08) 0.536 (0.93)			
	B: Net exp	posure, Eastern	Europe and Ch	ina separately		
	(1) Benchmark	(2) 1988-1998	(3) 1998-2008	(4) Final goods only		
Δ net exposure Eastern Europe Δ net exposure China	0.874*** (0.34) 0.080 (0.14)	1.005* (0.59) -0.268 (0.75)	1.131* (0.65) 0.098 (0.14)	1.377** (0.59) 0.237 (0.15)		

Table 3: Eastern Europe versus China: Baseline Results

Notes: N=739 (panel A and panel B column 1), N=326 (panel B column 2), N=413 (panel B column 3). Δ *import exposure* and Δ *export exposure* in panel A comprise only trade with Eastern Europe (column 1) and China (column 2), respectively. Δ *net exposure* in panel B is constructed by subtracting (1) from (2). See online appendix for the categorization of final goods. All regressions include the full set of control variables from the benchmark model of table 1, column 5. Standard errors clustered at the level of 50 aggregate labor market regions in parentheses. Levels of significance: *** 1 %, ** 5 %, * 10 %.

equal to 0.874, which is highly significant and substantially larger than the 0.237 estimated above for the joint "East". For China, on the other hand, the impact of net export exposure is estimated to be basically zero. The message is thus consistent with panel A of Table 3: the global effects of trade exposure with "the East" seem to be mostly driven by Eastern Europe.

One possible concern with this conclusion is that our estimation approach may not be equally well suited to capture the nature of German trade exposure towards the two different areas. Specifically, rising German trade with Eastern Europe could originate in genuinely German developments, rather than by an exogenous rise in productivity and market access of the Eastern European countries. In particular, trade could be mainly driven by a deeper fragmentation of German value chains, which in turn fosters offshoring to Eastern Europe, or by other sources of internal productivity growth in Germany. We proceed in three different ways to address this concern.

Separate time periods. First, in columns 2 and 3 of Table 3-B, we estimate the same model as in column 1 but now separately for the two time periods. We consider this to be an important exercise, because the period 1988–1998 captures the quasi-natural experiment of the sudden fall of the iron curtain, and the subsequent rise of the new market economies in Eastern Europe, much better than the period 1998–2008 does. Vice versa, if rising German trade with Eastern Europe were actually driven by German productivity growth or developments such as offshoring, we would expect this to be mainly relevant during the second time window, but not the first.

As can be seen in columns 2 and 3, our results for the two separate periods are similar as in the pooled model. For Eastern Europe, we estimate a positive and significant impact of net export exposure. Standard errors are a bit larger than in the pooled model, since we now have fewer observations, but the qualitative conclusions are the same. For China, we consistently find no effects in either time period.

Final versus intermediate goods. Second, the alternative story outlined above suggests that German trade with Eastern Europe could be mainly characterized by an export of intermediate goods that are processed abroad, and then returned to Germany for final assembly. To address this concern we use input-output tables to distinguish final from intermediate goods (see the online appendix for details). We then repeat our specification from column 1 of Table 3-B only for net export exposure in final goods, with the instrument adjusted accordingly. Results are shown in column 4.

We find that the results for Eastern Europe continue to hold, and that the marginal effect even gets a bit larger when we only consider final goods exposure. This is consistent with our estimation approach and interpretation of the baseline results, and less consistent with the alternative story about offshoring. Moreover, we still find no significant effects from net exposure with respect to China.

Cars. Third and finally, we address the special role of the automotive sector. As shown in Section 2.4, cars and car parts account for the biggest chunk of German trade with Eastern Europe. Even more importantly, in that sector we would especially expect that rising trade volumes could be driven by value chain fragmentation or productivity gains in Germany, rather than by the rise of Eastern Europe.

To investigate these issues, we re-estimate column 1 of Table 3-B but exclude cars and car parts. Results are reported in column 1 of Appendix Table (A.9). We find that the coefficient for net export exposure of Eastern Europe drops by about 10 %. Not surprisingly the standard error also increases, but the coefficient still remains significant. For China, the coefficient is virtually unchanged. In the further columns of Appendix Table (A.9), we respectively drop other sectors from the analysis. For those cases, results are very similar to the baseline findings. Hence, although cars are undoubtedly very important for Germany, they do not seem to drive all results.

Summing up, the results from this Section suggest that Eastern Europe has affected German labor markets more strongly than China. Moreover, and very importantly, we have provided several pieces of evidence which suggest that our instrumental variable approach seems well applicable to uncover causal effects in this context. Of course, we cannot rule out that some part of the rising trade with Eastern Europe may be due to genuinely German developments, such as productivity growth or unobservable shocks. However, our results suggest that at least a major chunk of the observed trade increase actually stems from the exogenous "rise" of productivity and accessibility of Eastern Europe, as posited by our empirical approach.

4.2 Why Eastern Europe? Why not China?

The question remains *why* the German labor market responds so strongly to the rise of Eastern Europe, and in the same vein, *why* there seem to be almost no displacement effects stemming from the rise of China. The latter conclusion is in stark contrast to *ADHs* findings for the US, and it would be hard to explain this difference solely by the fact that Germany runs a persistent overall trade surplus while the US runs a trade deficit. The German trade balance vis-a-vis China is also negative (see Figure 1), albeit to a smaller extent than the American one, but still the Chinese import exposure did not seem to cause major job losses. In this subsection we aim to provide a possible explanation for the differential impact of Eastern Europe and China in Germany.

Heterogeneous effects across industries. Recall that German local export exposures to Eastern Europe and China are highly correlated (0.835). Consequently, what should be driving the different overall impact of the two areas is the differential composition of the *import* exposures, where the correlation is only 0.328 (see Appendix Table A.4).

As in Section 3, we again aggregate the trade flows of Eastern Europe and China and consider the German net exports to the joint "East". However, we now split up sectors into two groups: the first group consists of those $X = \{10, 20, 30\}$ industries where imports from China, or respectively from Eastern Europe, increased the most between 1988 and 2008. The second group consists of all the other industries. We then repeat the specification from column 1 of Table 2, but now allow the two different groups of industries to have a different impact on manufacturing employment. Column 1 of Table 4 shows the estimation results.¹⁹

	Dependent variable: 10-year change manufacturing employment / working age pop. in %-points			
	(1)	(2)	(3)	
	Top 10	Top 20	Top 30	
	A: Eastern Europe importing industries			
Δ net exposure top Eastern Europe	0.298	0.211*	0.275**	
	(0.19)	(0.11)	(0.12)	
Δ net exposure all other industries	0.197*	0.295	0.162	
	(0.11)	(0.19)	(0.13)	
		B: China importing in	dustries	
Δ net exposure top China	0.119	0.147	0.161	
	(0.13)	(0.12)	(0.11)	
Δ net exposure all other industries	0.422**	0.503**	0.507**	
	(0.19)	(0.21)	(0.22)	

Table 4: Eastern Europe versus China: Heterogeneous effects across industries

Notes: N=739. Δ net exposure top Eastern Europe (China), is constructed using only the 10/20/30 industries where imports from Eastern Europe (China) grew the fastest between 1988 and 2008. All regressions include the full set of control variables from the benchmark model of table 1, column 5. Standard errors clustered at the level of 50 aggregate labor market regions in parentheses. Levels of significance: *** 1 %, ** 5 %, * 10 %.

¹⁹The instruments are constructed accordingly using the different net exports of the other countries.

Comparing these estimates to the "homogenous" industry coefficient of 0.237 from Table 2, we consistently find that the marginal effect of trade exposure is smaller for the top import industries from China than for the average industry. In other words, rising German trade in those labor-intensive sectors does not seem to have displaced many domestic jobs. For the top import industries from Eastern Europe, on the other hand, Table 4 shows that the marginal effect of trade exposure is above the average. That is, there are notable labor market effects from trade exposure in those industries where imports from Eastern Europe grew most rapidly (e.g., iron, steel, or car parts).

Initial trade balance and heterogeneous effects. To understand why this might be the case, it is instructive to consider Germany's initial trade balance in 1988. We consider the X = 30 sectors where imports from China grew the most between 1988 and 2008. We then correlate the sectoral changes in Chinese import values with the initial trade balance of Germany (with respect to the entire world) in 1988. This correlation is negative (-0.48) and statistically highly significant (p < 0.01). In other words, Germany tended to import goods from such sectors already in 1988 in which China *subsequently* became the world's most dominant supplier; the rise of China then tended to displace the imports from other countries.



Figure 2: German Imports of Textiles, 1998-2008

Consider the textile and apparel industry as an example. Figure 2 shows the evolution of German textile imports from China (1998-2008) and the joint volumes of textile imports from Greece, Italy and Turkey, which were previously the countries with the highest textile sales in the German market. The figure shows the rapid increase in textile imports from China, particularly since 2001, but at the same time an almost parallel reduction of the import volume from the other three countries.

The graph may thus provide at least a partial explanation for the weak labor market effects of Chinese import exposure: Germany imported textiles (and other goods where China developed a strong comparative advantage) already initially, so that domestic production and employment were already small to begin with. The "rise of China" then led to a re-direction of import flows, which supposedly caused job losses in the other countries, but not in Germany.²⁰

Importantly, for imports from Eastern Europe we obtain a very different picture. Considering the X = 30 sectors in which imports from Eastern Europe grew the most since 1988, we obtain a positive and highly significant correlation of +0.48 (p < 0.01) with Germany's initial trade balance. In other words, Germany tended to *export* goods in 1988 where the subsequent rise of Eastern Europe was particularly strong. This is then consistent with the pronounced job displacement effects of rising import penetration that we find in panel A of Table 3. Yet, bearing again in mind that most trade with Eastern Europe is intra-industry, those job losses were more than offset by the rising export opportunities in the same sectors for Germany.

We like to stress that we see the evidence presented in this sub-section as mostly suggestive, shedding some light on the differential impact of Eastern Europe and China in the German labor markets. Yet, this tentative evidence may also be helpful to develop an intuition for the stark differences between our results and the findings by *ADH* for the US. In particular, they provide evidence for substantial direct displacement of domestic American workers by Chinese imports. This corresponds well with the observation that rising Chinese imports did not lead to a comparably strong diversion of import flows away from other countries, but fuelled a large aggregate US trade deficit.

5 Other local labor market outcomes

In this section we consider the impact of the rise of the East on other outcome variables across German local labor markets. Specifically, we analyze regional population sizes, median wages, unemployment rates and non-manufacturing employment.

Population shifts. In their analysis, *ADH* emphasize sluggish adjustment of population across local labor markets. In other words, they find no evidence for population shifts in response to rising Chinese import competition in the US.

Returning to our benchmark specification from column 5 of Table 1, we address the impact of rising trade exposure on the change in (log) population sizes across German regions in column 1 of Table 5. Both for export and for import exposure, we obtain only very small effects, and moreover, these coefficients are very imprecisely estimated and statistically not distinguishable from zero.²¹ Hence, in line with *ADH*'s results, we also find that the local adjustments to trade shocks do not seem to involve major population shifts across regions. This is also consistent with the consensus view in the literature (see Molloy et al.; 2011) that regional labor mobility is even lower in Germany than

²⁰Large imports do not necessarily imply small domestic production of the respective sector, for example if trade is mostly intra-industrial in nature. However, with inter-industry trade such a negative relationship between imports and domestic production is indeed implied by standard trade theory.

²¹Notice that the dependent variable in column 1 is now a log change. The coefficients, thus, indicate that an increase in export (import) exposure by $1000 \in$ would increase (decrease) the regional population size by as little as 0.168 (0.165) per cent. The effects are, thus, very small and noisy.

in the US. The possibility that the rise of the East has caused at least small migratory responses is, however, not decisively rejected by the data since the estimated effects are very noisy. We will revisit this issue in our worker-level analysis in Section 6.

		Dependent variables: 10-year change						
	(1) working-age population	(2) median wage	(3) unemployment	(4) non-manufacturing employment	(5) total employment			
Δ import exposure Δ export exposure	-0.165 (0.14) 0.168	-0.016 (0.02) 0.108***	0.009 (0.02) -0.096*	-0.131* (0.08) 0.227*	-0.320** (0.13) 0.626***			
	(0.17)	(0.04)	(0.05)	(0.13)	(0.24)			

Table 5: Other labor market indicators

Notes: N=739. The dependent variables in column 1 is log change in working age population \times 100, and in column 2 the log change in the regional median wage \times 100. The dependent variables in columns 3–5 are, respectively, the percentage point change in the number unemployed persons/non-manufacturing workers/all workers in the local working age population. All regressions include the full set of control variables from the benchmark model of table 1, column 5. Standard errors clustered at the level of 50 aggregate labor market regions in parentheses. Levels of significance: *** 1%, ** 5%, * 10%.

Wages and unemployment. A detailed analysis how trade integration has affected wages and income inequality in Germany is beyond the scope of this paper. However, column 2 provides a brief look at such price responses. There we address the impact of export and import exposure on the change in (log) median wages across local labor markets. We find that an increase in local export exposure by $1,000 \in$ leads to a small increase of the regional wage level by about 0.1 per cent. Rising import exposure, on the other hand, does not seem to cause regional wage declines.

This corroborates our earlier finding that German labor markets have responded to rising trade exposure mainly at the employment margin, since induced wage responses are small at best. Moreover, there is some evidence for asymmetric effects, which could stem from downward nominal wage rigidity that is often found to be a feature of the German labor market (Dustmann et al. (2009)).

In column 3 we analyze the effects on the number of registered unemployed persons in the regional working-age populations. We obtain two main findings. First, more export-exposed regions experience decreasing unemployment rates, which is consistent with our previous results on net employment gains in those locations. Second, there is again an asymmetry since higher import exposure apparently does not cause higher unemployment rates, despite the wage rigidities discussed before. A possible explanation, apart from labor force exit, is that there are numerous active labor market policies in Germany which target workers who have been laid off or face a substantial risk thereof. These programs may cushion possible adverse import shocks, as workers prone to becoming unemployed are either retained in their original job with reduced hours (*Kurzarbeit*), or they may be quickly transferred into an active labor market measure in which case they are not counted as unemployed. **Non-manufacturing and total employment.** In column 4 of Table 5 we show that higher export exposure causes employment gains not only within the manufacturing sector, but also beyond. Specifically, we find a significantly positive effect on regional non-manufacturing employment relative to working age population. Analogously, higher import exposure causes job losses also in non-manufacturing sectors. Such "spillovers" may exist as demand from the manufacturing sector for local services is likely to increase (decrease) in regions with an export-oriented (import-competing) initial industry structure, thus causing employment adjustments also in those industries that are not (or, at least, not so much) directly affected by the rise of the East.

We further disentangle the employment reactions in different non-manufacturing industries in Appendix Table A.10. There we find that these spillover effects are almost exclusively concentrated on business services.²² For the other service branches, namely the construction sector and personal services, we find no such evidence.

Finally, column 5 reports the impact on total regional employment relative to population. Converting these marginal effects, and using a similar conservative benchmarking as in Section 3.2, the results imply that import penetration from the East has reduced the aggregate German employment rate by about 1.05 percentage points over the period 1988-2008. The rise in export exposure, by contrast, has led to an increase by about 1.91 percentage points. This corresponds to a *net* creation of around 442,000 full-time equivalent jobs that would not be there without the rise of the East, out of which around 69 % are within and 31 % are outside the manufacturing sector.

6 Worker-level evidence

In this final section, we complement our aggregate analysis with an investigation how trade liberalization has affected individual worker biographies. Specifically, we address the impact of rising import and export exposure from "the East" on expected job durations and, thus, on the stability of individual employment relationships.

6.1 Data and estimation approach

We use the Sample of Integrated Labour Market Biographies (SIAB). This data stems from all German social security notifications in the years 1975 to 2008. A two percent random sample has been drawn from all persons who have either been employed or officially registered as job-seekers, resulting in an individual-level spell data set with information on age, sex, nationality, qualification, occupation, spell durations, etc. This data is highly accurate even on a daily base due to its original purpose of calculating retirement pensions, and the previously used establishment-level data (the BHP) can

²²Upon closer inspection, we find evidence that within the business services, temporary work agencies ("*Zeitarbeit*") are responsible for a notable part of the overall effect. In this sector, agencies "rent" workers who perform production or management tasks to manufacturing establishments, which can flexibly adjust labor inputs without having to hire workers permanently. Our results suggest that export exposure has complementary effects on regular manufacturing jobs, and on these temporary jobs.

be merged to it. From the data, we identify all working-age individuals (age 22-64) who have been employed in manufacturing at the beginning of 1988 (respectively, 1998), and we then construct our dependent variable as the cumulative days in employment over the subsequent ten-year period.

To quantify how an individual worker is (potentially) affected by the rise of the East over the same period, we now construct the measures of trade exposure at the *industry* level. Specifically, the changes in import and export exposure per worker in a German industry *j* for $t = \{1988 - 1998, 1998 - 2008\}$ are defined as

$$\Delta(Import \ exp.)_{jt}^{EAST} = \frac{\Delta Im_{jt}^{D \leftarrow EAST}}{E_{jt}} \quad \text{and} \quad \Delta(Export \ exp.)_{jt}^{EAST} = \frac{\Delta Ex_{jt}^{D \rightarrow EAST}}{E_{jt}}$$

where $\Delta Im_{jt}^{D \leftarrow EAST}$ and $\Delta Ex_{jt}^{D \rightarrow EAST}$ are, respectively, the change in the German total import (export) volume from (to) China and Eastern Europe in industry *j* during the period *t* (in 1000 \in of 2005), and E_{jt} is total national industry employment at the beginning of the period. An overview of the data is provided in Appendix Table A.11.

For identification, we rely on a similar instrumental variable approach as in Autor, Dorn, Hanson and Song (2013). Specifically, for each person, we consider the sector of employment three years prior to the start of the observation period (denoted j - 3). We then measure the change in trade flows of the "instrument countries" vis-a-vis the East in the respective industry, divided by lagged national industry employment. This tackles possible issues of measurement error and sorting due to anticipation effects:

$$\Delta(Import \ exp._{Instr.})_{jt}^{EAST} = \frac{\Delta Im_{j-3t}^{Other \leftarrow EAST}}{E_{j-3t-3}} \ , \ \Delta(Export \ exp._{Instr.})_{jt}^{EAST} = \frac{\Delta Ex_{j-3t}^{Other \rightarrow EAST}}{E_{j-3t-3}}$$

In the regression, we furthermore use region \times time interaction terms, and we include dummies to control whether a worker was employed in the automotive sector or another tradable goods industry. Additionally, we use dummies for standard individual-level control variables, as well as year of birth dummies.

Since import and export exposure only vary across industries, one might worry that they capture industry-level effects that correlate with the change in trade exposure. To mitigate this problem, we also include further industry-level control variables in the regression, more specifically the Herfindahl-Index of establishment sizes, the Ellison and Glaeser (1997) agglomeration-index, the share of plants younger than two years, the average establishment size, the share of highly qualified employees, and the share of employees older than 50. Throughout, we allow our standard errors to be correlated between workers within the same industry and Federal State.

6.2 **Results and discussion**

The estimation results are reported in Table 6. Column 1 shows that a \leq 1,000 increase in import exposure per worker reduces the expected employment duration over 10 years by $0.313 \times \frac{365}{100} = 1.14$ days, ceteris paribus. Given that the average worker in

manufacturing has faced an increase of import exposure by about \in 19,280 over the period 1988-2008, this implies an overall decrease by about 28 days. By contrast, the average worker experienced a rise in export exposure by \in 23,298, which in turn is estimated to raise the expected employment duration by 105 days over 20 years.

In other words, rising import exposure from the East has increased the individual risk of job displacement for incumbent manufacturing workers in Germany. Yet, this is more than offset by a stabilizing effect stemming from the rising export opportunities. This finding complements our previous result that the rise of the East has, in the aggregate, created more manufacturing jobs than it has destroyed.

	Dependent variable: 100 x cumulative years of employment over 10 year period					
	2SLS	2SLS	2SLS	2SLS		
	(1)	(2)	(3)	(4)		
	total	plant	3-digit ind.	region		
Δ Imports	-0.313***	-1.105***	-1.032***	-0.658***		
per worker _{t=0}	(0.10)	(0.22)	(0.22)	(0.17)		
Δ Exports	1 236***	2.367***	2.265***	2 188***		
per worker $_{t=0}$ Female	(0.25) -167.714*** (2.58)	(0.78) -127.654***	(0.67) -142.822*** (3.49)	(0.41) -142.767*** (2.98)		
Foreign citizen	(2.58) -47.396*** (2.68)	-21.437*** (3.60)	(3.49) -29.092*** (3.55)	-31.238*** (3.10)		
Low Skilled High Skilled	-28.107*** (2.05) 35.945***	-13.625*** (2.94) -35.780***	-19.485*** (2.85) -14.306**	-9.535*** (2.52) -36.432***		
Employment in automotive sector in $t = 0$	(3.25)	(6.37)	(6.07)	(5.37)		
	-13.632	81.063	-44.130	-31.172		
	(20.28)	(50.21)	(43.28)	(30.89)		
Employment in other tradable	3.072	9.825	-19.337**	4.905		
goods industries in $t = 0$	(4.24)	(9.02)	(8.15)	(6.07)		
Industry level controls	Yes	Yes	Yes	Yes		
1st Stage F-statistic IP	220.177	220.177	220.177	220.177		
1st Stage F-statistic EP	38.202	38.202	38.202	38.202		

Table 6: Trade exposure and individual employment biographies

N=177,653. Standard errors clustered by 1,279 industry × Federal State cells, reported in parentheses. Control variables include dummy variables for start of period tenure, plant size, year of birth and Federal State × time period fixed effects. Models (2) – (4) consider cumulative employment only within the original establishment, 3-digit industry, and region, respectively. * $p \le 0.10$, ** $p \le 0.05$, *** $p \le 0.01$.

Our data allow us to dig deeper into these effects. In particular, the rising trade exposure from the East may have led individuals to change jobs across plants, industries, or regions without a notable unemployment spell, and these types of induced job churning remain invisible when we only consider overall employment durations.

Columns 2–4 in Table 6 show the impact of import and export exposure on expected tenure within the original plant, industry and region where the respective worker was initially employed. The results imply that the rising import exposure has causally increased job churning at all levels. In particular, the job duration in the original plant/industry/region was reduced by 78/73/46 days over 20 years for a worker who

faced the average increase in import competition. Yet, at the same time, rising export exposure has increased those durations by 201/193/186 days, respectively. The rise of the East has, therefore, stabilized individual work biographies on balance, as it has led to longer expected employment durations at the plant, industry and regional level for the average incumbent manufacturing worker.

In the remainder of this paper, we shed light on three specific questions that arise from these results: 1) how are work biographies of different types of individuals affected by the rise of the East, 2) how to relate our key finding that trade integration has stabilized jobs to recent theories of international trade with heterogeneous firms, and 3) how to square our worker-level findings with our earlier aggregate results, in particular with respect to trade-induced population shifts across regions.

Heterogeneous workers. In Appendix Table A.12 we split our sample of manufacturing workers, and consider how the effects of trade exposure on expected employment durations differ by the workers' education level, age, and gender.

The results suggest that young and old workers (age below/above the median) are affected similarly by the rise of the East. For male and female workers we also find, by and large, homogeneous effects. We do, however, observe notable differences how workers from different skill groups respond to trade shocks. In particular, while the rise of the East has led to an overall employment stabilization for all types of workers, we find that the low- and especially the medium-skilled are more responsive to trade exposure when it comes to job churning across plants and industries.

Medium-skilled workers have completed dual apprenticeship training, and their human capital is therefore relatively sector- and firm-specific. They are, thus, most vulnerable to rising import competition and face a higher individual displacement risk, but we still find that trade exposure has raised job stability also for this group.

Heterogeneous Establishments. In Appendix Table A.13 we investigate how trade exposure has affected workers differently, depending on the characteristics of the plant in which they were initially employed. Panels A and B distinguish individuals who originally worked for small/large plants (with size below/above the industry median within the same Federal State). Analogously, panels C and D distinguish whether the worker's original plant paid a median wage below/above the State-industry median.

We find that import exposure from the East has affected expected employment durations roughly similarly in all sub-samples. This true both for the overall employment duration, as well as for the plant-, industry- and region-specific tenures. In all cases, we find no significant differences in the estimated marginal effects between the different sub-samples of workers originally employed in small or large, or in "productive" (high-wage) or "less productive" (low-wage) establishments.

These empirical findings appear to be somewhat at odds with recent trade theories along the lines of Melitz (2003). From those models, one would expect that workers in small, less productive establishments should be hit harder by import shocks, but this prediction is not supported by our data. This is, however, consistent with recent results by Autor, Dorn, Hanson and Song (2013) who even find that import shocks in the US have affected employment durations in large establishments more severely than in small ones. More generally, a large literature in labor economics (von Wachter and Bender; 2006; Sullivan and von Wachter; 2009) documents that supply shocks seem to cause stronger displacement effects in large establishments.²³

With respect to export exposure, notice that we do not observe employment expansions or contractions at the establishment level on which the Melitz-model makes sharp predictions, but job durations of incumbent workers. Still, one would expect that the job stabilization effect should be stronger in expanding, exporting plants. Unfortunately, our data do not allow us to observe the plants' productivity or export status. Yet, our results from panel A and B of Appendix Table A.13 suggest that rising export exposure at the industry-level has prolonged job durations mostly in small establishments. For workers of large plants, we find much weaker (sometimes insignificant) effects, particularly when focussing on expected tenure with the original employer (see panels A and B). Although this also goes a bit against the intuition derived from the Melitz-model, it should be noted that the initial size of a plant is only a rough proxy for its productivity. This is especially true in the German context, where many small and medium-sized firms (*"Mittelstand"*) are actually highly active as exporters.²⁴

Distinguishing plants according to their initial median wage, as in panels C and D, is arguably a better proxy for their productivity and export status, even though Schank et al. (2007) find only a small wage premium for German exporters and Schmillen (2011) even points out that German exporters to new EU member states often pay no wage premium at all. Still, along those lines, we find that export exposure had a similar job stabilization effect in both sub-samples, and for plant- and industry-specific tenure this job stabilization was even slightly stronger for workers of "productive" establishments. This is better in line with standard theoretical reasoning.

Summing up, our empirical findings are by no means a refutation of recent trade models, but suggest that more work is needed to understand how rising trade exposure affects employment in heterogeneous worker-establishment matches as modeled, for example, in recent work by Helpman et al. (2012).

Trade-induced regional mobility. Finally, our results from column 5 of Table 6 indicate that rising trade exposure also affects the expected employment durations within the original region. In particular, rising export exposure raises region-specific tenure for a worker, while rising import exposure lowers it, and this pattern appears to be quite stable across different types of individuals (see Appendix Tables A.12 and A.13).

²³Our results, which suggest that large and small plants were affected similarly by import exposure, are thus even closer in spirit to the Melitz-model than those other studies.

²⁴In the recent framework by Holmes and Stevens (2012) there is also no monotonous relationship between firm size and export status. In their model, some firms produce standardized products while other firms (within the same industry) produce customized specialty goods, and the former firm type may be larger and more vulnerable to trade shocks than the latter.

How can we convert these coefficients into meaningful economic magnitudes? Consider the second time window (1998–2008). During this ten-year period, the average German manufacturing worker was employed for 7.9 years, and thereof 7.18 years in the original region where he or she was initially employed. This corresponds to an unconditional probability to leave the own region of about 9 per cent. Now, the average import shock from the East during this period was 13,188 \in , which in turn implies a reduction of region-specific tenure by 0.087 years, or an increase in the probability to move to another region by 1.1 percentage points. Correspondingly, the average export shock was 17,021 \in , which corresponds to a tenure prolongation of 0.372 years and a reduction of the migration probability by 4.7 percentage points. On average, we therefore find that rising trade exposure has *reduced* labor mobility in Germany.²⁵ This finding is thus consistent with our aggregate-level results from Table 5, which show that the rise of the East has not induced major population shifts across regions.

Still, considering only import exposure, we find negative effects on region-specific tenure for manufacturing workers. Generally speaking, even if there are such adjustments to trade shocks via individual mobility responses, this does in principle not jeopardize the applicability of our empirical approach which treats "regions" as small sub-units of the aggregate economy.²⁶ Quantitatively, however, we find that the implied magnitudes for such migratory responses are quite small, and moreover, we find little evidence that individual workers move systematically from import-exposed to more export-oriented regions.²⁷ Overall, we therefore conclude that our worker-level results are well in line with the insights from the aggregate-level analysis which suggested that the rise of the East has led to small population shifts at most.

7 Conclusions

The past decades have seen a strong increase in the volume of international trade. Deregulation and the abolishment of trade barriers as well as drastic reductions in transport costs have led to a steadily increasing integration of national economies. In this paper, we focus on two major facets of globalization: China's explosive ascent and the rise of Eastern Europe after the fall of the iron curtain. Understanding the consequences of those developments for the labor markets in the Western European

²⁵Descriptively, we observe a trend of increasing labor mobility in Germany over time. Appendix Table A.11 reports, for example, that overall employment durations have increased from the first to the second time period, while own-region tenure has decreased. Yet, our empirical findings suggest that the rise of the East was not the driving force behind this mobility increase.

²⁶The underlying theoretical framework of our empirical approach assumes immobile labor. However, that framework could potentially be generalized to allow for endogenous individual location decisions along the lines of the economic geography literature. Rising international trade exposure would then likely spur reallocations of workers across regions that are differently exposed to an aggregate trade shock, and this would "capitalize" in local wages and land prices in spatial equilibrium. Developing such a theoretical framework is left for future research, however.

²⁷To investigate this issue, we have constructed another tenure variable for each worker, namely the employment duration in regions with a greater export exposure than the original region. Using this as the dependent variable, we obtain only small and insignificant coefficients.

market economies is crucial, both from an economic and a political point of view.

We analyze the *causal* impact of the rise of China and Eastern Europe on the performance of local labor markets in Germany during the period 1988 to 2008, using an instrumental variable approach pioneered by Autor, Dorn and Hanson (2013). At the regional level, Germany is characterized by a substantial variation in local industry structures. These initial structures determine how the regions were affected by the rising trade exposure that kicked in since the mid 1990s.

The key message that can be derived from our analysis is that, overall, the rise in trade exposure has led to substantial employment gains in the German economy. But these gains are highly unevenly distributed across space. In fact, some regions have lost jobs as a result of the deeper trade integration, both in the manufacturing sector and beyond. But those losses were, in the aggregate, more than offset by additional jobs created in regions with industrial structures that allowed them to take advantage of the new export opportunities in the East. In our analysis at the individual level we complement this picture and show that trade exposure has, overall, led to more stable employment relationships by reducing the risk of job termination. However, trade again produces winners and losers, since workers in import competing industries indeed faced an increased risk of job churning and lower overall employment spells.

Our results for the German economy differ quite substantially from the findings for the United States. The US economy faced severe adverse effects on local labor markets due to the rapidly increasing Chinese import penetration, but the situation in Germany seems to be quite different. There we observe only negligible job displacement effects, and we provide a possible explanation why this was the case: Because the "rise of China" mainly diverted imports from other countries. Furthermore, our analysis suggests that focusing only on China provides an incomplete picture, at least in the German context. The rise of Eastern Europe had much stronger effects on German local labor markets. It caused substantial job losses, but even stronger job gains in regions that were specialized in the "right" types of industries. Furthermore, we show that the identification approach that was tailored to analyze the causal effects of the "rise of China" also seems applicable in the context of Eastern Europe.

An advantage of our approach is that it allows to analyze the local adjustments to trade exposure along many different margins. Our main focus on manufacturing employment is interesting, because in most industrialized countries there has been a long-run trend of structural change where employment secularly shifted away from the manufacturing sector and towards modern service industries. Our results suggest that trade with the East has per se decelerated this trend, and contributed to retaining the manufacturing sector in the German economy. Our conclusions for Germany may be representative for other developed economies to the extent that they also exhibit a notable specialization in modern, export-oriented manufacturing goods. An important avenue for future research would be to investigate if other countries are more similar to Germany or to the US in the way they adjust to rising trade exposure.

References

- Anderson, J. E. and van Wincoop, E. (2003). Gravity with gravitas: A solution to the border puzzle, *American Economic Review* **93**(1): 170–192.
- Autor, D. H., Dorn, D. and Hanson, G. H. (2013). The china syndrome: Local labor market effects of import competition in the united states, *American Economic Review* 103(6): 2121–68.
- Autor, D. H., Dorn, D., Hanson, G. H. and Song, J. (2013). Trade adjustment: Worker level evidence, NBER Working Paper 19226.
- Bertola, G. (2000). Labor markets in the european union, ifo Studien 1/2000, pp. 99-122.
- Blossfeld, H.-P. (1987). Labor market entry and the sexual segregation of careers in the federal republic of germany, *American Journal of Sociology* **93**(1): 89–118.
- Brülhart, M., Carrère, C. and Trionfetti, F. (2012). How wages and employment adjust to trade liberalization: Quasi-experimental evidence from austria, *Journal of International Economics* **86**(1): 68–81.
- Burda, M. C. and Severgnini, B. (2009). TFP Growth in Old and New Europe, *Comparative Economic Studies* **51**(4): 447–466.
- Chiquiar, D. (2008). Globalization, regional wage differentials, and the stolpersamuelson theorem: Evidence from mexico, *Journal of International Economics* **74**(1): 70–93.
- Dustmann, C., Ludsteck, J. and Schönberg, U. (2009). Revisiting the german wage structure, *Quarterly Journal of Economics* **124**(2): 843–881.
- Ebenstein, A., Harrison, A., McMillan, M. and Phillips, S. (2013). Estimating the impact of trade and offshoring on american workers using the current population surveys, *Review of Economics and Statistics*. forthcoming.
- Eckey, H.-F., Kosfeld, R. and Tuerck, M. (2006). Abgrenzung deutscher arbeitsmarktregionen, *Raumforschung und Raumordnung* **64**(4): 299–309.
- Ellison, G. and Glaeser, E. L. (1997). Geographic concentration in u.s. manufacturing industries: A dartboard approach, *Journal of Political Economy* **105**(5): 889–927.
- Feenstra, R. C. and Hanson, G. H. (1999). The impact of outsourcing and hightechnology capital on wages: Estimates for the u.s., 1979-1990, *Quarterly Journal of Economics* 114(3): 907–940.
- Feenstra, R. C. and Wei, S.-J. (2010). *China's Growing Role in World Trade*, NBER and University of Chicago Press.

- Glitz, A. (2012). The labor market impact of immigration: A quasi-experiment exploiting immigrant location rules in germany, *Journal of Labor Economics* **30**(1): pp. 175–213.
- Grossman, G. M. and Rossi-Hansberg, E. (2008). Trading tasks: A simple theory of offshoring, *American Economic Review* **98**(5): 1978–1997.
- Harrigan, J. (2000). International trade and american wages in general equilibrium, 1967-1995, *in* R. C. Feenstra (ed.), *The Impact of International Trade on Wages*, University of Chicago Press, pp. 171 196.
- Harrison, A., McLaren, J. and McMillan, M. (2010). Recent findings on trade and inequality, NBER Working Paper No. 16425.
- Helpman, E., Itskhoki, O., Muendler, M.-A. and Redding, S. J. (2012). Trade and inequality: From theory to estimation, *Technical report*, National Bureau of Economic Research.
- Holmes, T. J. and Stevens, J. J. (2012). An alternative theory of the plant size distribution, with geography and intra-and international trade, *Journal of Political Economy* **forthcoming**.
- Kovak, B. K. (2011). Local labor market effects of trade policy: Evidence from brazilian liberalization, mimeo, Carnegie Mellon University.
- Kropp, P. and Schwengler, B. (2011). Abgrenzung von arbeitsmarktregionen * ein methodenvorschlag, *Raumforschung und Raumordnung* **69**(1): 45–62.
- Melitz, M. J. (2003). The impact of trade on intra-industry reallocations and aggregate industry productivity, *Econometrica* .
- Molloy, R., Smith, C. L. and Wozniak, A. K. (2011). Internal migration in the united states, *Journal of Economic Perspectives* **25**(2): 1–42.
- Redding, S. J. and Sturm, D. M. (2008). The costs of remoteness: Evidence from german division and reunification, *American Economic Review* **98**(5): 1766–1797.
- Schank, T., Schnabel, C. and Wagner, J. (2007). Do exporters really pay higher wages? first evidence from german linked employer-employee data, *Journal of International Economics* **72**(1): 52–74.
- Schmillen, A. (2011). The exporter wage premium reconsidered destinations, distances and linked employer-employee data, OEI Working Paper No. 305.
- Spengler, A. (2008). The establishment history panel, *Schmollers Jahrbuch Journal of Applied Social Science Studies* **128**(3): 501–509.

- Sullivan, D. and von Wachter, T. (2009). Job displacement and mortality: An analysis using administrative data, *Quarterly Journal of Economics* **124**(3): 1265–1306.
- Topalova, P. (2010). Factor immobility and regional impacts of trade liberalization: Evidence on poverty from india, *American Economic Journal: Applied Economics* **2**(4): 1–41.
- von Wachter, T. and Bender, S. (2006). In the right place at the wrong time: The role of firms and luck in young workers' careers, *American Economic Review* **96**(5): 1679–1705.

Supplementary online appendix – not for publication

The Rise of the East and the Far East: German Labor Markets and Trade Integration

Wolfgang Dauth

Sebastian Findeisen

Jens Suedekum

December 2013

A) Some background about trade integration and productivity growth in Eastern Europe

While the different aspects of the "rise of China" are well documented in *ADH*'s work and elsewhere, comparatively less is known about the trade integration and productivity trends in Eastern Europe after the fall of the iron curtain. In this appendix we provide a very brief overview.

Trade integration. The following time line summarizes the major policy steps towards trade integration of the 21 Eastern European countries considered in our study.



As can be seen, Eastern European trade integration is a continuous and still ongoing process. Still, for many countries relatively close to, and important for Germany, no-table events occurred early during the years 1995 and 1996, i.e., in the first ten-year period considered in our empirical analysis. During those years, several countries joined the WTO or signed partnership arrangements. Further countries followed shortly afterwards. The next major step then occurred in 2004, during our second ten-year time period, when several Eastern European countries joined the European Union (Bulgaria and Romania followed in 2007).

A quantitative measure for the magnitude of trade barriers between Germany and Eastern Europe is difficult to obtain, but the time line strongly suggests that such barriers have been falling significantly during the observation period of our study, form almost infinitely high levels prior to 1989.

Productivity growth. We also briefly document the trend of Eastern European productivity growth during the observation period. The following figure depicts average labor productivity (total output over hours worked in manufacturing) across the 21 Eastern European countries (2010=100). Data stem from national accounts data as compiled by EUROSTAT and refer to the time period 1995–2008, as data coverage is insufficient for the earlier years. For comparison, the figure also includes the German productivity trend (2010=100) over the same period.



The figure clearly shows a differential trend of stronger productivity growth in Eastern Europe relative to Germany, and thus a catching-up of "the East" relative to "the West" over these years. This conclusion is also supported in the study by Burda and Severgnini (2009), who consider various methods to deal with severe measurement error present in the output and productivity data of the transition economies in Central and Eastern Europe. They find high and increasing rates of TFP growth in those economies over the period 1994-2009, outpacing the corresponding trends in "old Europe" (including Germany).

Summing up, the "rise of Eastern Europe" seems to have many similar features as the "rise of China", especially during the first decade after the fall of the iron curtain: Relatively strong productivity growth and gains in accessibility stemming from a massive reduction in trade barriers vis-a-vis Western Europe and Germany, in particular.

B) Trade exposure on third markets

In columns 1 and 2 of Table 2 we first consider the standard local measure of net export exposure that is obtained by subtracting eq. (1) from eq.(2):

$$\Delta(Net \ export \ exp.)_{it}^{EAST} = \Delta(Export \ exp.)_{it}^{EAST} - \Delta(Import \ exp.)_{it}^{EAST} \\ = \sum_{j} \frac{E_{ijt}}{E_{jt}} \frac{\Delta E x_{jt}^{D \to EAST} - \Delta I m_{jt}^{D \leftarrow EAST}}{E_{it}}, \quad (A.1)$$

where $\Delta E x_{jt}^{D \to EAST}$ and $\Delta I m_{jt}^{D \leftarrow EAST}$ are the changes in aggregate German exports/ imports to/from "the East" in industry j between *t* and *t* + 1. In column 1 of Table 2 we correspondingly use the net exports of Australia, Canada, Japan, Norway, New Zealand, Sweden, Singapore, and the United Kingdom vis-a-vis "the East" (China and Eastern Europe) as an instrument for (A.1).

The import exposure part of this measure is "domestic" in the sense that it measures how German producers from region *i* are displaced by the rise of the East across their domestic (German) markets. As an alternative to eq.(1), we now consider an extended measure of import exposure along the lines of what *ADH* do in their Table 10, Panel B. Specifically, we take into account that German producers are also displaced in other markets, namely in foreign third-countries, and compute

$$\Delta (Import \ exp.)_{it}^{EAST,O} = \sum_{j} \frac{E_{ijt}}{E_{jt}} \frac{\Delta Im_{jt}^{D \leftarrow EAST} + \sum_{O} \frac{X_{jt}^{D \rightarrow O}}{X_{jt}^{D \rightarrow world}} \Delta Im_{jt}^{O \leftarrow EAST}}{E_{it}}$$
(A.2)

In contrast to (1), this measure includes the change in imports of other countries, denoted as *O*, from the East and weights the importance of the different other countries by their share in total German exports in the respective industry. As "other countries" we use all Western European EU members.

We then compute *net export exposure* analogously as in (A.1) by subtracting (A.2) from (2): $\Delta(Net \ export \ exp.)_{it}^{EAST,O} = \Delta(Export \ exp.)_{it}^{EAST} - \Delta(Import \ exp.)_{it}^{EAST,O}$. That is, we use the same measure for local export exposure as before but exchange the measure for import exposure. The instrument for this new measure $\Delta(Net \ export \ exp.)_{it}^{EAST,O}$ is constructed analogously, by using the trade flows of the "instrument countries" to/from the East and the other Western European EU members. Column 3 of Table 2 then reports the 2SLS coefficient for the impact of this novel measure on manufacturing employment relative to working-age population across German regions.

As a further robustness check, we have also considered an extended measure for local *export* exposure. Specifically, analogously to (A.2), we compute:

$$\Delta(Export \ exp.)_{it}^{EAST,O} = \sum_{j} \frac{E_{ijt}}{E_{jt}} \frac{\Delta E x_{jt}^{D \to EAST} + \sum_{O} \frac{M_{jt}^{D \leftarrow O}}{M_{jt}^{D \leftarrow world}} \Delta E x_{jt}^{O \to EAST}}{E_{it}}$$
(A.3)

This measure captures the idea that the rise of the Eastern markets also diverts ex-

port flows of the other Western EU countries towards "the East", which in turn leads to tougher competition for German exporters there. The local measure for *net export exposure* is then calculated by now subtracting (A.2) from (A.3), with the instrument adjusted accordingly.

Using that measure it turns out, however, that the estimation results are virtually identical to those in column 3 of Table 2. In particular, we estimate a marginal effect on manufacturing employment that is still equal to 0.205, and remains highly significant. In other words, third-market effects seem to play a more important role for local import exposure than for local export exposure.

C) The identification approach based on gravity residuals

In this appendix, we describe the alternative identification approach that is based on a gravity approach to international trade. Consider a standard gravity equation as in the seminal paper by Anderson and van Wincoop (2003):

$$X_{ik}^{j} = \frac{y_{i}^{j} y_{k}^{j}}{Y_{W}^{j}} \cdot \left(\frac{\tau_{ik}}{P_{i}^{j} P_{k}^{j}}\right)^{1-\sigma}$$
(A.4)

Here, X_{ik}^j is the total export flow from country *i* to country *k* in industry *j* (time subscripts are omitted for now). World expenditure on the goods from that industry is denoted as Y_W^j , and country *i*'s expenditure level is y_i^j . The variable P_i^j denotes the standard CES price index in *i* for *j*, σ is the elasticity of substitution, and τ_{ik}^j are iceberg trade costs between countries *i* and *k*.

We use this gravity equation to express total (log) exports from "the East" (i = E) to some destination country k, and analogously, from Germany (i = D) to k as follows:

$$\log X_{Ek}^{j} = \log \left[\frac{1}{Y_{W}^{j}} \cdot \frac{y_{k}^{j}}{(P_{k}^{j})^{1-\sigma}} \right] + \log \left[\frac{y_{E}^{j}}{(P_{E}^{j})^{1-\sigma}} \right] - (\sigma - 1) \log \tau_{Ek}^{j}$$
$$\log X_{Dk}^{j} = \log \left[\frac{1}{Y_{W}^{j}} \cdot \frac{y_{k}^{j}}{(P_{k}^{j})^{1-\sigma}} \right] + \log \left[\frac{y_{D}^{j}}{(P_{D}^{j})^{1-\sigma}} \right] - (\sigma - 1) \log \tau_{Dk}^{j}$$

Hence, East relative to German exports to country k can be written as

$$\log X_{Ek}^{j} - \log X_{Dk}^{j} = \left(\log z_{E}^{j} - \log z_{D}^{j}\right) - (\sigma - 1) \left[\log \tau_{Ek}^{j} - \log \tau_{Dk}^{j}\right], \quad (A.5)$$

where $z_i^j = y_i^j / (P_i^j)^{1-\sigma}$. The term $(\log z_E^j - \log z_D^j)$ thus captures the East's comparative advantage in industry j (relative to Germany), which includes relative productivity, domestic expenditure, toughness of competition (measured by P_i^j), and so on. The term $(\log \tau_{Ek}^j - \log \tau_{Dk}^j)$ captures relative accessibility of market k from the East and Germany, respectively.

To implement the gravity approach, we estimate the following specification:

$$\log X_{Ekt}^j - \log X_{Dkt}^j = \alpha_j + \alpha_k + \epsilon_{jkt}$$
(A.6)

That is, we regress East relative to German sectoral exports to various high-income countries k, pooled over the period 1988-2008, on industry and destination market fixed effects. We recover the residual from this regression. Using (A.5), this residual has the following interpretation:

$$\epsilon_{jkt} = \left[\log \left(\frac{z_{Et}^j}{z_{Dt}^j} \right) - \alpha_j \right] + \left[-(\sigma - 1) \log \left(\frac{\tau_{Ekt}^j}{\tau_{Dkt}^j} \right) - \alpha_k \right].$$
(A.7)

It thus captures East's productivity and trade cost advantage vis-a-vis Germany in industry j and destination market k in year t. Taking the difference between time period t and t - 1 then gives the East's increase in relative competitiveness vis-a-vis Germany in market k as follows:

$$\Delta \epsilon_{jkt} = \epsilon_{jkt} - \epsilon_{jk(t-1)} = \log\left(\frac{z_{Et}^j / z_{Dt}^j}{z_{E(t-1)}^j / z_{D(t-1)}^j}\right) - (\sigma - 1)\log\left(\frac{\tau_{Ekt}^j / \tau_{Dkt}^j}{\tau_{Ek(t-1)}^j / \tau_{Dk(t-1)}^j}\right),$$

and taking the average over the N_k different markets k yields

$$\Delta \bar{\epsilon}_{jt} = \log\left(\frac{z_{Et}^j}{z_{E(t-1)}^j} \cdot \frac{z_{D(t-1)}^j}{z_{Dt}^j}\right) - \frac{(\sigma-1)}{N_k} \sum_k \log\left(\frac{\tau_{Ekt}^j}{\tau_{Ek(t-1)}^j} \cdot \frac{\tau_{Dk(t-1)}^j}{\tau_{Dkt}^j}\right)$$
(A.8)

The term $\Delta \overline{\epsilon}_{jt}$ can be interpreted as the average increase in East's competitiveness relative to Germany, as recovered from relative export flows to various high-income third markets. Exponentiating this term,

$$\widetilde{\Delta \overline{\epsilon}}_{jt} = \exp^{\Delta \overline{\epsilon}_{jt}} = \frac{\left(\frac{z_{Et}^j/z_{Dt}^j}{z_{E(t-1)}^j/z_{D(t-1)}^j}\right)}{\prod_{k=1}^{N_k} \left(\frac{\tau_{Ekt}^j/\tau_{Dkt}^j}{\tau_{Ek(t-1)}^j/\tau_{Dk(t-1)}^j}\right)^{\frac{\sigma-1}{N_k}}}$$

we can introduce a different measure for regional import exposure in Germany, using this implied gain in relative competitiveness computed from gravity residuals:

$$\Delta IPW_{it}^{grav} = \sum_{j} \frac{E_{ij(t-1)}}{E_{j(t-1)}} \cdot \frac{\overline{\Delta \overline{\epsilon_{jt}}} \cdot M_{t-1}^{j}}{E_{i(t-1)}}, \qquad (A.9)$$

That is, instead of the actual (observed) changes in sectoral imports from the East as in eq. (1) in the main text, we use a hypothetical increase in imports, constructed from the initial import *level* (M_{t-1}^j) multiplied by the implied increase in relative competitiveness of the East on other markets.

Gravity measure for export exposure One of the contributions of our approach is to consider exports to the East. Our argument is that, due to the rise of that area, German producers are not only displaced in their home markets and in other foreign markets. Germany can also export to the East, because rising income there also implies higher demand for German products. This aspect – the rise of new markets for German exporters – is missing in the gravity exposure measure A.8, because export flows from Germany to the East are neglected there.

To tackle this issue, we develop an analogous gravity measure for local export exposure in Germany. This new measure captures the differential rise of the attractiveness of the East as a destination market for Germany, relative to other possible destinations to which Germany may export. Starting from the same gravity equation as above, we can write

$$\log X_{DE}^{j} = \log \left[\frac{1}{Y_{W}^{j}} \cdot \frac{y_{D}^{j}}{(P_{D}^{j})^{1-\sigma}} \right] + \log \left[\frac{y_{E}^{j}}{(P_{E}^{j})^{1-\sigma}} \right] - (\sigma - 1) \log \tau_{DE}^{j}$$
$$\log X_{Dk}^{j} = \log \left[\frac{1}{Y_{W}^{j}} \cdot \frac{y_{D}^{j}}{(P_{D}^{j})^{1-\sigma}} \right] + \log \left[\frac{y_{k}^{j}}{(P_{k}^{j})^{1-\sigma}} \right] - (\sigma - 1) \log \tau_{Dk}^{j}$$

Hence, German industry j exports to the East relative to some other country k are

$$\log X_{DE}^{j} - \log X_{Dk}^{j} = \left(\log z_{E}^{j} - \log z_{k}^{j}\right) - (\sigma - 1) \left[\log \tau_{DE}^{j} - \log \tau_{Dk}^{j}\right],$$
(A.10)

Estimating a similar gravity equation as before, where relative German (log) exports to the East and to other countries k are regressed on fixed effects for industry j (denoted α_j^{μ}) and destination market k (denoted α_k^{μ}), the residual now has the following interpretation:

$$\mu_{jkt} = \left[\log \left(\frac{z_{Et}^j}{z_{kt}^j} \right) - \alpha_j^{\mu} \right] + \left[-(\sigma - 1) \log \left(\frac{\tau_{DEt}^j}{\tau_{Dkt}^j} \right) - \alpha_k^{\mu} \right]$$
(A.11)

Again taking the difference of μ_{jkt} over a ten-year period, and then taking the average across the *k* different other countries, we have

$$\Delta \overline{\mu_{jt}} = \frac{1}{N_k} \left(\sum_k \log \left(\frac{z_{Et}^j}{z_{E(t-1)}^j} \cdot \frac{z_{k(t-1)}^j}{z_{kt}^j} \right) \right) - \frac{(\sigma - 1)}{N_k} \left(\sum_k \log \left(\frac{\tau_{DEt}^j}{\tau_{DE(t-1)}^j} \cdot \frac{\tau_{Dk(t-1)}^j}{\tau_{Dkt}^j} \right) \right)$$
(A.12)

The measure A.12 thus captures the increase of the East's attractiveness from a German perspective, relative to the "average other country". This gain in attractiveness can be driven either by a differential increase in income/expenditure, or by a relatively strong decline in trade costs from Germany to the East. Exponentiating, $\widetilde{\Delta \mu_{jt}} = \exp^{\Delta \mu_{jt}}$, and multiplying it with the initial level of German exports , we hence

obtain a gravity-based measure for local export exposure:

$$\Delta EPW_{it}^{grav} = \sum_{j} \frac{E_{ij(t-1)}}{E_{j(t-1)}} \cdot \frac{\widetilde{\Delta \mu_{jt}} \cdot X_{t-1}^{j}}{E_{i(t-1)}}, \qquad (A.13)$$

as opposed to the raw measure of export exposure from eq. (2) in the main text that uses the observed increases in German exports.

Implementing a gravity-based measure for net export exposure Using (A.9) and (A.13), we derive the following theory-based gravity measure for net export exposure:

$$\Delta NET_{it}^{grav} = \sum_{j} \frac{E_{ij(t-1)}}{E_{j(t-1)}E_{i(t-1)}} \cdot X_{t-1}^{j} \cdot \underbrace{\left(\widetilde{\Delta \mu}_{jt} - \widetilde{\Delta \overline{\epsilon}}_{jt} \cdot \frac{M_{t-1}^{j}}{\underbrace{X_{t-1}^{j}}_{=\rho_{t-1}^{j}}\right)}_{=\Delta \Xi_{it}^{grav}}$$
(A.14)

In (A.14), ρ_0^j indicates whether Germany was a net exporter ($\rho_{t-1}^j < 1$) or a net importer ($\rho_{t-1}^j > 1$) of good j in the initial period. Using (A.8) and (A.12), the industry-specific term in parentheses can be expressed in the following way:

$$\Delta \Xi_{jt}^{grav} = (1 - \rho_{t-1}^{j}) \log \left[\frac{z_{Et}^{j}}{z_{E(t-1)}^{j}} \right] + \rho_{t-1}^{j} \log \left[\frac{z_{Dt}^{j}}{z_{D(t-1)}^{j}} \right] - \frac{1}{N_{k}} \sum_{k} \log \left[\frac{z_{kt}^{j}}{z_{k(t-1)}^{j}} \right] - (\sigma - 1) \log \left[\frac{\tau_{DEt}^{j}}{\tau_{DE(t-1)}^{j}} \right] + \rho_{t-1}^{j} \frac{(\sigma - 1)}{N_{k}} \sum_{k} \log \left[\frac{\tau_{Ekt}^{j}}{\tau_{Ek(t-1)}^{j}} \right] - (1 - \rho_{t-1}^{j}) \frac{(\sigma - 1)}{N_{k}} \sum_{k} \log \left[\frac{\tau_{Dkt}^{j}}{\tau_{Dk(t-1)}^{j}} \right]$$
(A.15)

This expression in (A.15) neatly shows how different exogenous changes affect German net exports of industry *j*-goods in general equilibrium. First, and most importantly, an improvement of East's competitiveness in industry j ($z_{Et}^j/z_{E(t-1)}^j > 1$) raises net German exports, ceteris paribus, only if $\rho_{t-1}^j < 1$, i.e., if Germany was a net exporter of this good. Otherwise, if $\rho_{t-1}^j > 1$, the improvement of the East lowers net German exports. This highlights the double-edged effect of the "rise of the East" on German trade: Import exposure rises as Eastern firms become more competitive, while exports exposure rises as consumers in the East become richer and demand more German goods. Which effect dominates is ambiguous, a priori. Moreover, decreasing access costs from Germany to the East ($\tau_{DEt}^j/\tau_{DE(t-1)}^j < 1$) clearly improve German net exports, while an improvement of the East's access to other countries hurts German net exports, but less so the higher is the net export level (the smaller is ρ_{t-1}^j).

In subsection (3.2), we then perform an OLS estimation using this gravity-based measure for local trade exposure from (A.15), see column 4 of Table 2 . As destination

markets k we again use the Western European EU members, but results hardly change if we additionally add the US and the "instrument countries" to that list. Notice that we can estimate eq.(3) by OLS, since the gravity approach by construction identifies the rise of East's competitiveness and accessibility, and sterilizes the impacts of possible confounding shocks.

D) Final versus intermediate goods

In Section 4 we distinguish final and intermediate goods, see column 4 of Table 3-B. To distinguish between these two goods categories, we adopt a similar approach as *ADH*, see their Table 10, panel C. We use the 1998 input-output table provided by the German Federal Statistical Office. At the level of 72 products that can be matched to our industry classification, this table provides information on use of a good's production or imports. We use the information on the shares of World imports that are used for consumption or investment rather than as an input by any industry. We multiply these shares with the German imports from Eastern Europe or China. This gives us a measure on the share of each sector's imports in final goods.

We then follow an analogous approach to separate exports of intermediate versus final goods, but now use only information on domestic production that is not exported. This distinction should work reasonably well if two assumptions hold: first, the shares of inputs and final goods in imports from Eastern Europe and China should be similar to the corresponding shares in total imports. Second, German goods should be similarly divided into final and intermediate goods in Eastern Europe and China as they are in domestic production and consumption.

That information then allows us to net total trade from intermediate goods in order to arrive at a measure for final goods trade exposure.



Figure A.1: Change in import exposure 1998-2008



Figure A.2: Change in export exposure 1998-2008

Appendix Tables

	Industry	2008	1998	1988					
Imp	Imports from Eastern Europe								
341	Manuf. of motor vehicles	7100	4440	76					
343	Manuf. of parts and accessories for motor vehicles and their engines	6830	1610	11					
274	Manuf. of basic precious and non-ferrous metals	4280	1940	992					
271	Manuf. of basic iron and steel and of ferro-alloys (ECSC1)	3510	949	402					
316	Manuf. of electrical equipment n.e.c.	3350	1260	26					
Expo	orts to Eastern Europe								
$34\bar{1}$	Manuf. of motor vehicles	13300	3970	248					
343	Manuf. of parts and accessories for motor vehicles and their engines	9180	2610	92					
295	Manuf. of other special purpose machinery	7830	3400	1250					
291	Manuf. of machinery for the production and use of mechanical power,	5390	1500	413					
	except aircraft, vehicle and cycle engines								
252	Manuf. of plastic products	5280	2090	577					

Table A.2: Trade volumes with China, top-5 manufacturing sectors

	Industry	2008	1998	1988				
Imp	Imports from China							
300	Manuf. of office machinery and computers	8630	1160	12				
182	Manuf. of other wearing apparel and accessories	4950	1900	704				
365	Manuf. of games and toys	3280	658	46				
323	Manuf. of television and radio receivers, sound or video recording or	2930	700	171				
	reproducing apparatus and associated goods							
321	Manuf. of electronic valves and tubes and other electronic components	2920	123	2				
Expo	orts to China							
$34\overline{1}$	Manuf. of motor vehicles	3530	238	209				
295	Manuf. of other special purpose machinery	3220	1050	590				
291	Manuf. of machinery for the production and use of mechanical power,	2740	248	108				
	except aircraft, vehicle and cycle engines							
294	Manuf. of machine-tools	1900	376	306				
343	Manuf. of parts and accessories for motor vehicles and their engines	1640	114	31				
Industry 1	evel trade volumes measured in Million Euros of 2005. Sorted by trade volume in 2008.							

	1988-1998		199	8-2008	
	Dependent Variable				
Δ manuf. employment / working age pop.	-2.507	(2.707)	-0.150	(2.213)	
	Trade exposure				
Δ imports per worker _{t=0}					
China	0.592	(0.522)	1.903	(1.885)	
Eastern Europe	1.804	(1.001)	1.848	(1.299)	
Both	2.396	(1.322)	3.751	(2.649)	
Δ exports per worker _{t=0}					
China	0.134	(0.110)	1.037	(0.816)	
Eastern Europe	2.174	(1.009)	3.714	(2.269)	
Both	2.308	(1.055)	4.752	(3.003)	

Table A.3: Descriptive statistics of the main variables at the regional level

Notes: Mean values and standard deviations (in parentheses). Dependent variable in %-points.

Trade exposure variables constructed according to equations 1 and 2 in 1000 Euros of 2005 per worker.

Table A.4: Correlations of regional import and export exposure measures for China and Eastern Europe trade

		Chi	ina	Eastern Europe		
		import exp.	export exp.	import exp.	import exp.	
China	import exp.	1				
China	export exp.	0.3945	1			
Eastern Europe	import exp.	0.3281	0.5032	1		
Eastern Europe	export exp.	0.4559	0.8354	0.7490	1	

Notes: Correlation coefficients.

	manufact	De uring	ependent var employmer	iable: 10-yo it / workin	ear change g age pop. in %-points
Panel A: Change in Regional A	ggregatior	ı			
	Dro	p Eas	t Germany	La	abor Market Regions
	(1)	(2)	(3)) (4)
Δ import exposure	-0.255***	-0.2	29***	-0.234	4*** -0.212***
	(0.08)	(0	.07)	(0.0	8) (0.08)
Δ export exposure	0.428**	0.4	18**	0.367	0.380***
	(0.21)	(0	.21)	(0.1	2) (0.11)
Region \times time	Yes		_	Ye	s –
State × time	-	Y	les	-	Yes
First stage results, dependent va	riable: Δ is	mpor	t exposure		
Δ import exposure	0.236***	0.23	38***	0.284	.291***
(other countries)	(0.06)	(0	.06)	(0.0)	5) (0.05)
Δ export exposure	0.115***	0.1	15***	0.12	1* 0.128*
(other countries)	(0.04)	(0	.04)	(0.0)	(0.07)
K-squared	0.759	15	167	0.84	6 0.857
F-lest of excluded listfullents	10.037	15	.100	20.0	47 10.740
First stage results, dependent va	riable: $\Delta \epsilon$	export	t exposure		
Δ import exposure	0.003	0.	004	0.02	20 0.025
(other countries)	(0.02)	(0)	.02)	(0.0)	(0.02)
Δ export exposure	(0.07)	0.40	07)	0.352	(0.07)
(other countries)	(0.07)	(0)	.07) 827	(0.0	(0.07) (0.07)
F-test of excluded instruments	21 410	21	280	14.8	64 14 087
	21.110			11.0	
Panel B: Change in Instrum	ent Group	,			
_	Ove	er-	Leave out	Add	Add DK, NL, BE,
	identi	ified	UK	USA	LU, FR, CH, AT
	(5)	(6)	(7)	(8)
Δ import exposure	-0.14	2***	-0.191***	-0.200**	-0.173**
	(0.0	5)	(0.07)	(0.08)	(0.07)
Δ export exposure	0.40	6**	0.397*	0.366*	0.328*
	(0.1	6)	(0.22)	(0.22)	(0.17)
First stage results, dependen	t variable:	Δ im	iport exposu	re	
Δ import exposure			0.301***	0.096***	0.163***
(other countries)			(0.08)	(0.03)	(0.04)
Δ export exposure			0.108***	0.092***	0.050*
(other countries)			(0.03)	(0.03)	(0.03)
R-squared	0.78	39	0.749	0.721	0.759
F-test of excluded instrumen	ts 53.4	12	20.025	9.807	9.447
First stage results, dependen	t variable:	Δex	port exposu	re	
Δ import exposure			-0.002	-0.000	-0.016
(other countries)			(0.02)	(0.01)	(0.01)
Δ export exposure			0.393***	0.282***	0.325***
(other countries)	0.00	20	(0.08)	(0.05)	(0.03)
K-squared	0.89	99 150	0.812	0.812	0.854
F-test of excluded instrumen	ts 123.2	239	16.984	19.193	48.867

Table A.5: Robustness Checks – Regional classification and instrument group

Notes: N=652 (column 1 and 2), N=259 (column 3 and 4), and N=739 (columns 5–8). In panel B, the instruments are adjusted to entail the reported countries. All regressions include the full set of control variables from the benchmark model of table 1, column 5. Standard errors clustered at the level of 50 aggregate labor market regions in parentheses. Levels of significance: *** 1 %, ** 5 %, * 10 %.

		Depende	nt variable: 1()-vear change		
	manufact	uring empl	oyment / wor	king age pop. in %-points		
	(1)	(2)	(3)	(4)		
Δ import exposure	-0.158**	-0.160**	-0.190***	-0.180***		
1 1	(0.07)	(0.06)	(0.07)	(0.07)		
Δ export exposure	0.425**	0.426**	0.399*	0.394*		
1 1	(0.21)	(0.20)	(0.21)	(0.21)		
% manuf. of other tradable goods	-0.073***	-0.078***	-0.068***	-0.073***		
0	(0.02)	(0.02)	(0.02)	(0.02)		
% manuf. of cars	-0.081**	-0.083**	-0.074**	-0.076**		
	(0.03)	(0.03)	(0.03)	(0.03)		
% high skilled	-0.042	-0.065	-0.039	-0.065		
C	(0.04)	(0.04)	(0.04)	(0.05)		
% foreigners	-0.159***	-0.161***	-0.162***	-0.167***		
0	(0.04)	(0.04)	(0.04)	(0.04)		
% women	-0.061***	-0.060***	-0.060***	-0.058***		
	(0.01)	(0.01)	(0.01)	(0.01)		
% routine occupations	-0.021	-0.029	-0.014	-0.023		
	(0.03)	(0.03)	(0.03)	(0.03)		
Time dummy	Yes	Yes	_	_		
Region dummies	Region	State	_	_		
Interactions	-	-	Region \times t	State \times t		
First stage results, dependent variable: Δ import exposure						
Δ import exposure	0.281***	0.281***	0.252***	0.253***		
(other countries)	(0.06)	(0.06)	(0.06)	(0.06)		
Δ export exposure	0.153***	0.154***	0.105***	0.105***		
(other countries)	(0.03)	(0.03)	(0.03)	(0.03)		
R-squared	0.822	0.824	0.763	0.772		
F-test of excluded instruments	47.499	47.059	16.361	15.358		
First stage results, dependent varia	ble: $\Delta \exp ($	ort exposure	e			
Δ import exposure	0.043*	0.043*	0.002	0.003		
(other countries)	(0.02)	(0.02)	(0.01)	(0.02)		
Δ export exposure	0.484***	0.486***	0.393***	0.391***		
(other countries)	(0.06)	(0.06)	(0.06)	(0.06)		
R-squared	0.852	0.853	0.826	0.832		
F-test of excluded instruments	40.773	40.451	22.280	21.862		

Table A.6: Robustness check: Federal State Dummies

Notes: N=739. All regressions include the full set of control variables from the benchmark model of table 1, column 5. Standard errors clustered at the level of 50 aggregate labor market regions in parentheses. Levels of significance: *** 1 %, ** 5 %, * 10 %.

	Depende manufacturing empl	ent variable: 10-year change oyment / working age pop. in %-points
	(1) Eastern Europe	(2) China
Δ import exposure	-0.973***	-0.149*
	(0.34)	(0.08)
Δ export exposure	0.900***	0.536
	(0.32)	(0.93)
% manuf. of other tradable goods	-0.070**	-0.046**
C C	(0.03)	(0.02)
% manuf. of cars	-0.070*	-0.036
	(0.04)	(0.03)
% high skilled	-0.012	-0.051
0	(0.05)	(0.04)
% foreigners	-0.168***	-0.155***
0	(0.03)	(0.03)
% women	-0.059***	-0.060***
	(0.01)	(0.01)
% routine occupations	0.012	-0.025
	(0.04)	(0.03)
First stage results, dependent variable	e: Δ import exposure	
Δ import exposure	0.563***	0.234***
(other countries)	(0.12)	(0.06)
Δ export exposure	-0.057	0.041
(other countries)	(0.04)	(0.04)
R-squared	0.655	0.703
F-test of excluded instruments	11.172	9.399
First stage results, dependent variable	e: Δ export exposure	
Δ import exposure	0.352***	-0.012***
(other countries)	(0.12)	(0.00)
Δ export exposure	0.508***	0.103***
(other countries)	(0.04)	(0.02)
R-squared	0.804	0.712
F-test of excluded instruments	93.149	12.256

Table A.7: Eastern Europe versus China (I): Import and export exposure separately

Notes: N=739. Δ *import exposure* and Δ *export exposure* comprise only trade with Eastern Europe (column 1) and China (column 2), respectively. All regressions include the full set of control variables from the benchmark model of table 1, column 5. Standard errors clustered at the level of 50 aggregate labor market regions in parentheses. Levels of significance: *** 1 %, ** 5 %, * 10 %.

	De manufacturing	ependent vari 3 employmen	able: 10-year cł t / working age	nange 2 pop. in %-points
	(1) Benchmark	(2) 1988-1998	(3) 1998-2008	(4) Final goods only
Δ net exposure EE	0.874***	1.005*	1.131*	1.377**
*	(0.34)	(0.59)	(0.65)	(0.59)
Δ net exposure CN	0.080	-0.268	0.098	0.237
*	(0.14)	(0.75)	(0.14)	(0.15)
% manuf. of other tradable goods	-0.072***	-0.092***	-0.066	-0.059***
C C	(0.02)	(0.02)	(0.05)	(0.02)
% manuf. of cars	-0.077***	-0.063**	-0.127	-0.045**
	(0.02)	(0.03)	(0.10)	(0.02)
% high skilled	-0.015	-0.131	-0.033	-0.021
0	(0.05)	(0.08)	(0.06)	(0.05)
% foreigners	-0.169***	-0.171***	-0.114***	-0.164***
-	(0.03)	(0.05)	(0.04)	(0.04)
% women	-0.059***	-0.086***	-0.021	-0.059***
	(0.01)	(0.02)	(0.02)	(0.01)
% routine occupations	0.011	-0.019	0.020	0.016
_	(0.04)	(0.04)	(0.06)	(0.04)
First stage results, dependent varia	ble: Δ net expos	ure EE		
Δ net exposure EE	0.488***	0.483***	0.275**	0.474***
(other countries)	(0.07)	(0.16)	(0.11)	(0.10)
Δ net exposure CN	-0.005	0.098***	-0.008	-0.015
(other countries)	(0.01)	(0.03)	(0.02)	(0.01)
R-squared	0.670	0.363	0.738	0.544
F-test of excluded instruments	30.833	14.089	5.511	14.322
First stage results, dependent varia	ble: Δ net expos	ure CN		
Δ net exposure EE	0.101	0.021	-0.237**	0.012
(other countries)	(0.07)	(0.05)	(0.11)	(0.09)
Δ net exposure CN	0.218***	0.145***	0.252***	0.242***
(other countries)	(0.05)	(0.01)	(0.06)	(0.06)
R-squared	0.509	0.711	0.521	0.557
F-test of excluded instruments	11.851	49.216	8.281	9.427

Table A.8: Eastern Europe versus China (II): Net exposure, Eastern Europe and China separately

Notes: N=739 (column 1), N=326 (column 2), N=413 (column 3). Δ *net exposure* is constructed by subtracting (1) from (2). See online appendix for the categorization of final goods. All regressions include the full set of control variables from the benchmark model of table 1, column 5. Standard errors clustered at the level of 50 aggregate labor market regions in parentheses. Levels of significance: *** 1 %, ** 5 %, * 10 %.

		manufactu	Dependent variable: 10-year char ring employment / working age p	ıge 2p. in %-points	
Omitted industry	Automotive	Spec. purp. machinery	Mach. for prod. of mech. power	Basic chemicals	Gen. purp. machinery
Δ net exposure EE	0.778*	0.864***	0.849**	0.760**	0.853**
٩	(0.45)	(0.33)	(0.33)	(0.32)	(0.34)
Δ net exposure CN	0.082	0.101	0.073	0.185	0.084
4	(0.13)	(0.14)	(0.14)	(0.11)	(0.14)
First stage results, dependent ve	ariable: Δ net expo	sure EE			
Δ net exposure EE	0.324***	0.487***	0.492***	0.498^{***}	0.488^{***}
(other countries)	(0.0)	(0.07)	(0.02)	(0.07)	(0.07)
Δ net exposure CN	-0.001	-0.00	-0.009	-0.012	-0.005
(other countries)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
R-squared	0.554	0.650	0.661	0.668	0.663
F-test of excluded instruments	9.938	29.200	30.108	30.924	30.691
First stage results, dependent v	ariable: Δ net expo	sure CN			
Δ net exposure EE	-0.099	0.096	0.106	0.089	0.100
(other countries)	(0.0)	(0.07)	(0.07)	(0.07)	(0.07)
Δ net exposure CN	0.227***	0.216^{***}	0.213***	0.224^{***}	0.218^{***}
(other countries)	(0.05)	(0.05)	(0.05)	(0.05)	(0.05)
R-squared	0.499	0.509	0.525	0.513	0.509
F-test of excluded instruments	10.340	11.479	11.852	11.382	11.871
Omitted industry	Office machines	Wearing apparel	Communication devices	Electrical equipment	Furniture
Δ net exposure EE	0.890^{***}	0.865***	0.917***	0.872**	0.825***
ł	(0.33)	(0.32)	(0.33)	(0.34)	(0.32)
Δ net exposure CN	0.059	0.042	0.104	0.061	0.109
4	(0.20)	(0.14)	(0.14)	(0.12)	(0.12)
First stage results, dependent ve	ariable: Δ net expo	sure EE			
Δ net exposure EE	0.487***	0.499***	0.492***	0.488^{***}	0.498^{***}
(other countries)	(0.07)	(0.07)	(0.02)	(0.07)	(0.07)
Δ net exposure CN	0.005	-0.003	-0.004	-0.001	-0.008
(other countries)	(0.02)	(0.01)	(0.01)	(0.01)	(0.01)
R-squared	0.670	0.629	0.675	0.672	0.681
F-test of excluded instruments	34.683	35.795	29.821	32.124	32.568
First stage results, dependent ve	ariable: Δ net expo	sure CN			
Δ net exposure EE	0.115^{*}	0.102	0.089	0.125*	0.091
(other countries)	(0.07)	(0.07)	(0.07)	(0.07)	(0.07)
Δ net exposure CN	0.225***	0.214^{***}	0.214^{***}	0.241^{***}	0.226***
(other countries)	(0.05)	(0.05)	(0.05)	(0.05)	(0.05)
R-squared F-test of excluded instruments	0.534 17 253	0.487 11 944	0.484 9.236	0.586	0.516 11.383
Notes: N=739. All trade exposure variables and instru	iments are constructed by om	tting the respective industries. All regression	ns include the full set of control variables from the bench	mark model of table 1. column 5. Sta	ndard errors clustered at the level of 50

Table A.9: Robustness checks: Drop most important industries for trade with Eastern Europe and China

ivouss: iv=/>2. Au trace exposure variables and instruments are constructed by omitting the respective industries. aggregate labor market regions in parentheses. Levels of significance: *** 1 %, ** 5 %, * 10 %.

	Dependen	t variables: 10-yea / working age po	ar change in employment p. in %-points
	cons-	personal	business
	truction	services	services
Δ import exposure	0.008	-0.059	-0.050
	(0.01)	(0.04)	(0.04)
Δ export exposure	0.027	-0.012	0.188**
	(0.02)	(0.05)	(0.09)

Table A.10: Impact on non-manufa	cturing	industries
----------------------------------	---------	------------

Notes: N=739. The dependent variables are constructed analogously to the dependent variable of the benchmark model as the percentage point change of the number of workers in each of the denoted subsectors relative to the working age population. All regressions include the full set of control variables from the benchmark model of table 1, column 5. Standard errors clustered at the level of 50 aggregate labor market regions in parentheses. Levels of significance: *** 1 %, ** 5 %, * 10 %.

		1988-1998	199	8-2008
		Dependent	variables	
Cumulative years of employment:				
total	7.637	(2.957)	7.886	(2.837)
in original establishment	6.174	(3.679)	5.516	(3.898)
in original 3-digit industry	6.444	(3.627)	6.142	(3.815)
in original region	7.219	(3.231)	7.176	(3.327)
		Tue de sur		
		Irade exp	osure	
Δ imports per worker _{t=0}				
Eastern Europe	4.689	(4.719)	6.610	(9.411)
China	1.404	(3.590)	6.578	(20.171)
Both	6.092	(6.797)	13.188	(23.170)
Δ exports per worker _{t=0}				
Eastern Europe	5.882	(5.391)	13.163	(10.816)
China	0.395	(0.980)	3.858	(4.408)
Both	6.277	(5.835)	17.021	(13.538)

Table A.11: Descriptive statistics of the main variables at the worker level

Mean values and standard deviations (in parentheses) for manufacturing workers. Cumulative years of employment based on employment data with daily precision. Trade exposure measured in \in 1,000 per worker

	100 x cum	De ulative vear	pendent variable	e: over 10 vear pe
			2010	
	25LS (1)	25LS (2)	25L5 (3)	25LS (4)
	total	plant	3-digit ind.	region
	A	: Low Skille	ed	
Δ Imports	-0.369*	-0.708**	-0.712**	-0.499**
per worker $_{t=0}$	(0.20)	(0.28)	(0.28)	(0.22)
Δ Exports	1.212***	1.550**	1.275*	1.601***
per worker $_{t=0}$	(0.43)	(0.78)	(0.69)	(0.51)
N	42179	42179	42179	42179
1st Stage F-statistic IP	95.884	95.884	95.884	95.884
1st Stage F-statistic EP	25.849	25.849	25.849	25.849
	B: 1	Medium Ski	lled	
Δ Imports	-0.307***	-1.262***	-1.143***	-0.805***
per worker $_{t=0}$	(0.10)	(0.22)	(0.21)	(0.19)
Δ Exports	1.105***	2.593***	2.458***	2.383***
per worker $_{t=0}$	(0.26)	(0.76)	(0.66)	(0.47)
N	116202	116202	116202	116202
lst Stage F-statistic IP	256.148	256.148	256.148	256.148
1st Stage F-statistic EP	38.479	38.479	38.479	38.479
	C	: High Skill	ed	
Δ Imports	-0.301**	-1.055**	-0.993**	-0.521
per worker $_{t=0}$	(0.15)	(0.47)	(0.48)	(0.38)
Δ Exports	1.211**	0.383	2.198	1.924*
per worker $_{t=0}$	(0.60)	(2.68)	(1.82)	(1.11)
N	11476	11476	11476	11476
lst Stage F-statistic IP	143.292	143.292	143.292	143.292
lst Stage F-statistic EP	25.068	25.068	25.068	25.068
	D: `	Younger tha	n 38	
Δ Imports	-0.354***	-1.101***	-1.096***	-0.743***
per worker $_{t=0}$	(0.12)	(0.24)	(0.26)	(0.19)
Δ Exports	1.461***	2.489***	2.239***	2.344***
per worker $_{t=0}$	(0.25)	(0.78)	(0.69)	(0.45)
N	88714	88714	88714	88714
1st Stage F-statistic IP	175.149	175.149	175.149	175.149
lst Stage F-statistic EP	36.485	36.485	36.485	36.485
	E:	Older than	38	
Δ Imports	-0.282**	-1.117***	-0.987***	-0.599***
per worker $_{t=0}$	(0.12)	(0.25)	(0.21)	(0.20)
Δ Exports	1.022***	2.152***	2.223***	2.001***
per worker $_{t=0}$	(0.34)	(0.81)	(0.69)	(0.47)
N	88939	88939	88939	88939
Ist Stage F-statistic IP	249.733	249.733	249.733	249.733
lst Stage F-statistic EP	37.828	37.828	37.828	37.828
		F: Female		
Δ Imports	-0.366**	-1.002***	-0.842***	-0.652***
per worker $_{t=0}$	(0.17)	(0.29)	(0.27)	(0.21)
Δ Exports	0.986**	2.086**	1.326	1.731***
per worker $_{t=0}$	(0.44)	(0.89)	(0.85)	(0.57)
N	48000	48000	48000	48000
lst Stage F-statistic IP	166.369	166.369	166.369	166.369
lst Stage F-statistic EP	42.950	42.950	42.950	42.950
		G: Male		
Δ Imports	-0.244***	-1.117***	-1.034***	-0.587***
per worker $_{t=0}$	(0.09)	(0.21)	(0.22)	(0.21)
Δ Exports	1.321***	2.450***	2.524***	2.327***
per worker $_{t=0}$	(0.26)	(0.81)	(0.66)	(0.45)
N	129653	129653	129653	129653
st Stage F-statistic IP	258.596	258.596	258.596	258.596
st Stage F-statistic EP	28.769	28.769	28.769	28.769

Table A.12: Trade exposure and individual employment biographies

Control variables include dummy variables for start of period tenure, plant size, year of birth and region \times time interactions. Models (2) – (4) consider cumulative employment only within the original establishment, 3-digit industry, and region, respectively. Standard errors clustered by 1,279 industry \times federal state cells in parentheses. * p \leq 0.10, ** p \leq 0.01

		Dr	nondont variable	
	100 x cum	ulative year	s of employment	over 10 year period
	2SLS	2SLS	2SLS	2SLS
	(1)	(2)	(3)	(4)
	total	plant	3-digit ind.	region
A: Plant la	rger than sar	nple media	n of same industr	y/state
Δ Imports	-0.248**	-1.171***	-1.078***	-0.743***
per worker $_{t=0}$	(0.12)	(0.39)	(0.31)	(0.25)
Δ Exports	0.744**	1.359	1.801*	1.891***
per worker $_{t=0}$	(0.31)	(1.12)	(0.94)	(0.47)
N	88181	88181	88181	88181
1st Stage F-statistic IP	237.705	237.705	237.705	237.705
1st Stage F-statistic EP	57.110	57.110	57.110	57.110
B: Plant sn	naller than s	ample Medi	an same industry	/state
Δ Imports	-0.393***	-1.126***	-1.052***	-0.633***
per worker $_{t=0}$	(0.13)	(0.23)	(0.21)	(0.20)
Δ Exports	1.765***	3.450***	2.872***	2.666***
per worker $_{t=0}$	(0.33)	(0.76)	(0.65)	(0.58)
N N	89472	89472	89472	89472
1st Stage F-statistic IP	366.998	366.998	366.998	366.998
1st Stage F-statistic EP	68.879	68.879	68.879	68.879
C: Plant median was	ge higher tha	an populatio	on median of sam	e industry/state
Δ Imports	-0.327***	-1.197***	-1.133***	-0.657**
per worker $_{t=0}$	(0.11)	(0.25)	(0.31)	(0.29)
Δ Exports	1.037***	2.534**	2.660***	1.982***
per worker $_{t=0}$	(0.29)	(1.01)	(0.77)	(0.53)
Ň	93203	93203	93203	93203
1st Stage F-statistic IP	263.107	263.107	263.107	263.107
1st Stage F-statistic EP	33.686	33.686	33.686	33.686
D: Plant median wa	ge lower tha	n populatic	n median of same	e industry/state
Δ Imports	-0.313**	-0.926***	-0.878***	-0.642***
per worker $_{t=0}$	(0.14)	(0.31)	(0.23)	(0.18)
Δ Exports	1.452***	2.039**	1.767**	2.347***
per worker $_{t=0}$	(0.33)	(1.02)	(0.83)	(0.53)
Ň	84450	84450	84450	84450
1st Stage F-statistic IP	147.394	147.394	147.394	147.394
1st Stage F-statistic EP	34.900	34.900	34.900	34.900

Table A.13: Trade exposure and individual employment biographies

Control variables include dummy variables for start of period tenure, plant size, year of birth and region \times time interactions. Models (2) – (4) consider cumulative employment only within the original establishment, 3-digit industry, and region, respectively. Standard errors clustered by 1,279 industry \times federal state cells in parentheses. * $p \le 0.10$, ** $p \le 0.05$, *** $p \le 0.01$

PREVIOUS DISCUSSION PAPERS

- 128 Hottenrott; Hanna and Rexhäuser, Sascha, Policy-Induced Environmental Technology and Inventive Efforts: Is There a Crowding Out?, December 2013.
- 127 Dauth, Wolfgang, Findeisen, Sebastian and Suedekum, Jens, The Rise of the East and the Far East: German Labor Markets and Trade Integration, December 2013.
- 126 Wenzel, Tobias, Consumer Myopia, Competition and the Incentives to Unshroud Add-on Information, December 2013. Forthcoming in: Journal of Economic Behavior and Organization.
- 125 Schwarz, Christian and Suedekum, Jens, Global Sourcing of Complex Production Processes, December 2013. Forthcoming in: Journal of International Economics.
- 124 Defever, Fabrice and Suedekum, Jens, Financial Liberalization and the Relationship-Specificity of Exports, December 2013.
- 123 Bauernschuster, Stefan, Falck, Oliver, Heblich, Stephan and Suedekum, Jens, Why Are Educated and Risk-Loving Persons More Mobile Across Regions?, December 2013. Forthcoming in: Journal of Economic Behavior and Organization.
- 122 Hottenrott, Hanna and Lopes-Bento, Cindy, Quantity or Quality? Knowledge Alliances and their Effects on Patenting, December 2013.
- 121 Hottenrott, Hanna and Lopes-Bento, Cindy, (International) R&D collaboration and SMEs: The effectiveness of targeted public R&D support schemes, December 2013.
- 120 Giesen, Kristian and Suedekum, Jens, City Age and City Size, November 2013.
- 119 Trax, Michaela, Brunow, Stephan and Suedekum, Jens, Cultural Diversity and Plant-Level Productivity, November 2013.
- 118 Manasakis, Constantine and Vlassis, Minas, Downstream Mode of Competition With Upstream Market Power, November 2013.
- 117 Sapi, Geza and Suleymanova, Irina, Consumer Flexibility, Data Quality and Targeted Pricing, November 2013.
- 116 Hinloopen, Jeroen, Müller, Wieland and Normann, Hans-Theo, Output Commitment Through Product Bundling: Experimental Evidence, November 2013. Published in: European Economic Review 65 (2014), pp. 164-180.
- 115 Baumann, Florian, Denter, Philipp and Friehe Tim, Hide or Show? Endogenous Observability of Private Precautions Against Crime When Property Value is Private Information, November 2013.
- 114 Fan, Ying, Kühn, Kai-Uwe and Lafontaine, Francine, Financial Constraints and Moral Hazard: The Case of Franchising, November 2013.
- 113 Aguzzoni, Luca, Argentesi, Elena, Buccirossi, Paolo, Ciari, Lorenzo, Duso, Tomaso, Tognoni, Massimo and Vitale, Cristiana, They Played the Merger Game: A Retrospective Analysis in the UK Videogames Market, October 2013.

- 112 Myrseth, Kristian Ove R., Riener, Gerhard and Wollbrant, Conny, Tangible Temptation in the Social Dilemma: Cash, Cooperation, and Self-Control, October 2013.
- 111 Hasnas, Irina, Lambertini, Luca and Palestini, Arsen, Open Innovation in a Dynamic Cournot Duopoly, October 2013.
- 110 Baumann, Florian and Friehe, Tim, Competitive Pressure and Corporate Crime, September 2013.
- 109 Böckers, Veit, Haucap, Justus and Heimeshoff, Ulrich, Benefits of an Integrated European Electricity Market, September 2013.
- 108 Normann, Hans-Theo and Tan, Elaine S., Effects of Different Cartel Policies: Evidence from the German Power-Cable Industry, September 2013. Forthcoming in: Industrial and Corporate Change.
- 107 Haucap, Justus, Heimeshoff, Ulrich, Klein, Gordon J., Rickert, Dennis and Wey, Christian, Bargaining Power in Manufacturer-Retailer Relationships, September 2013.
- 106 Baumann, Florian and Friehe, Tim, Design Standards and Technology Adoption: Welfare Effects of Increasing Environmental Fines when the Number of Firms is Endogenous, September 2013.
- 105 Jeitschko, Thomas D., NYSE Changing Hands: Antitrust and Attempted Acquisitions of an Erstwhile Monopoly, August 2013.
- 104 Böckers, Veit, Giessing, Leonie and Rösch, Jürgen, The Green Game Changer: An Empirical Assessment of the Effects of Wind and Solar Power on the Merit Order, August 2013.
- 103 Haucap, Justus and Muck, Johannes, What Drives the Relevance and Reputation of Economics Journals? An Update from a Survey among Economists, August 2013.
- 102 Jovanovic, Dragan and Wey, Christian, Passive Partial Ownership, Sneaky Takeovers, and Merger Control, August 2013.
- 101 Haucap, Justus, Heimeshoff, Ulrich, Klein, Gordon J., Rickert, Dennis and Wey, Christian, Inter-Format Competition Among Retailers – The Role of Private Label Products in Market Delineation, August 2013.
- 100 Normann, Hans-Theo, Requate, Till and Waichman, Israel, Do Short-Term Laboratory Experiments Provide Valid Descriptions of Long-Term Economic Interactions? A Study of Cournot Markets, July 2013. Forthcoming in: Experimental Economics.
- 99 Dertwinkel-Kalt, Markus, Haucap, Justus and Wey, Christian, Input Price Discrimination (Bans), Entry and Welfare, June 2013.
- 98 Aguzzoni, Luca, Argentesi, Elena, Ciari, Lorenzo, Duso, Tomaso and Tognoni, Massimo, Ex-post Merger Evaluation in the UK Retail Market for Books, June 2013.
- 97 Caprice, Stéphane and von Schlippenbach, Vanessa, One-Stop Shopping as a Cause of Slotting Fees: A Rent-Shifting Mechanism, May 2012. Published in: Journal of Economics and Management Strategy, 22 (2013), pp. 468-487.
- 96 Wenzel, Tobias, Independent Service Operators in ATM Markets, June 2013. Forthcoming in: Scottish Journal of Political Economy.

- 95 Coublucq, Daniel, Econometric Analysis of Productivity with Measurement Error: Empirical Application to the US Railroad Industry, June 2013.
- 94 Coublucq, Daniel, Demand Estimation with Selection Bias: A Dynamic Game Approach with an Application to the US Railroad Industry, June 2013.
- 93 Baumann, Florian and Friehe, Tim, Status Concerns as a Motive for Crime?, April 2013.
- 92 Jeitschko, Thomas D. and Zhang, Nanyun, Adverse Effects of Patent Pooling on Product Development and Commercialization, April 2013.
- 91 Baumann, Florian and Friehe, Tim, Private Protection Against Crime when Property Value is Private Information, April 2013. Published in: International Review of Law and Economics, 35 (2013), pp. 73-79.
- Baumann, Florian and Friehe, Tim, Cheap Talk About the Detection Probability, April 2013.
 Forthcoming in: International Game Theory Review.
- 89 Pagel, Beatrice and Wey, Christian, How to Counter Union Power? Equilibrium Mergers in International Oligopoly, April 2013.
- Jovanovic, Dragan, Mergers, Managerial Incentives, and Efficiencies, April 2013.
- 87 Heimeshoff, Ulrich and Klein Gordon J., Bargaining Power and Local Heroes, March 2013.
- 86 Bertschek, Irene, Cerquera, Daniel and Klein, Gordon J., More Bits More Bucks? Measuring the Impact of Broadband Internet on Firm Performance, February 2013. Forthcoming in: Information Economics and Policy.
- Rasch, Alexander and Wenzel, Tobias, Piracy in a Two-Sided Software Market, February 2013.
 Published in: Journal of Economic Behavior & Organization, 88 (2013), pp. 78-89.
- 84 Bataille, Marc and Steinmetz, Alexander, Intermodal Competition on Some Routes in Transportation Networks: The Case of Inter Urban Buses and Railways, January 2013.
- 83 Haucap, Justus and Heimeshoff, Ulrich, Google, Facebook, Amazon, eBay: Is the Internet Driving Competition or Market Monopolization?, January 2013. Forthcoming in: International Economics and Economic Policy.
- 82 Regner, Tobias and Riener, Gerhard, Voluntary Payments, Privacy and Social Pressure on the Internet: A Natural Field Experiment, December 2012.
- 81 Dertwinkel-Kalt, Markus and Wey, Christian, The Effects of Remedies on Merger Activity in Oligopoly, December 2012.
- 80 Baumann, Florian and Friehe, Tim, Optimal Damages Multipliers in Oligopolistic Markets, December 2012.
- 79 Duso, Tomaso, Röller, Lars-Hendrik and Seldeslachts, Jo, Collusion through Joint R&D: An Empirical Assessment, December 2012. Forthcoming in: The Review of Economics and Statistics.
- Baumann, Florian and Heine, Klaus, Innovation, Tort Law, and Competition, December 2012.
 Forthcoming in: Journal of Institutional and Theoretical Economics.

- 77 Coenen, Michael and Jovanovic, Dragan, Investment Behavior in a Constrained Dictator Game, November 2012.
- 76 Gu, Yiquan and Wenzel, Tobias, Strategic Obfuscation and Consumer Protection Policy in Financial Markets: Theory and Experimental Evidence, November 2012. Forthcoming in: Journal of Industrial Economics under the title "Strategic Obfuscation and Consumer Protection Policy".
- 75 Haucap, Justus, Heimeshoff, Ulrich and Jovanovic, Dragan, Competition in Germany's Minute Reserve Power Market: An Econometric Analysis, November 2012. Forthcoming in: The Energy Journal.
- 74 Normann, Hans-Theo, Rösch, Jürgen and Schultz, Luis Manuel, Do Buyer Groups Facilitate Collusion?, November 2012.
- 73 Riener, Gerhard and Wiederhold, Simon, Heterogeneous Treatment Effects in Groups, November 2012.
- 72 Berlemann, Michael and Haucap, Justus, Which Factors Drive the Decision to Boycott and Opt Out of Research Rankings? A Note, November 2012.
- 71 Muck, Johannes and Heimeshoff, Ulrich, First Mover Advantages in Mobile Telecommunications: Evidence from OECD Countries, October 2012.
- 70 Karaçuka, Mehmet, Çatik, A. Nazif and Haucap, Justus, Consumer Choice and Local Network Effects in Mobile Telecommunications in Turkey, October 2012. Published in: Telecommunications Policy, 37 (2013), pp. 334-344.
- 69 Clemens, Georg and Rau, Holger A., Rebels without a Clue? Experimental Evidence on Partial Cartels, April 2013 (First Version October 2012).
- 68 Regner, Tobias and Riener, Gerhard, Motivational Cherry Picking, September 2012.
- 67 Fonseca, Miguel A. and Normann, Hans-Theo, Excess Capacity and Pricing in Bertrand-Edgeworth Markets: Experimental Evidence, September 2012. Published in: Journal of Institutional and Theoretical Economics, 169 (2013), pp. 199-228.
- 66 Riener, Gerhard and Wiederhold, Simon, Team Building and Hidden Costs of Control, September 2012.
- Fonseca, Miguel A. and Normann, Hans-Theo, Explicit vs. Tacit Collusion The Impact of Communication in Oligopoly Experiments, August 2012.
 Published in: European Economic Review, 56 (2012), pp. 1759-1772.
- 64 Jovanovic, Dragan and Wey, Christian, An Equilibrium Analysis of Efficiency Gains from Mergers, July 2012.
- 63 Dewenter, Ralf, Jaschinski, Thomas and Kuchinke, Björn A., Hospital Market Concentration and Discrimination of Patients, July 2012.
- 62 Von Schlippenbach, Vanessa and Teichmann, Isabel, The Strategic Use of Private Quality Standards in Food Supply Chains, May 2012. Published in: American Journal of Agricultural Economics, 94 (2012), pp. 1189-1201.
- 61 Sapi, Geza, Bargaining, Vertical Mergers and Entry, July 2012.
- Jentzsch, Nicola, Sapi, Geza and Suleymanova, Irina, Targeted Pricing and Customer Data Sharing Among Rivals, July 2012.
 Published in: International Journal of Industrial Organization, 31 (2013), pp. 131-144.

- 59 Lambarraa, Fatima and Riener, Gerhard, On the Norms of Charitable Giving in Islam: A Field Experiment, June 2012.
- 58 Duso, Tomaso, Gugler, Klaus and Szücs, Florian, An Empirical Assessment of the 2004 EU Merger Policy Reform, June 2012. Published in: Economic Journal, 123 (2013), F596-F619.
- 57 Dewenter, Ralf and Heimeshoff, Ulrich, More Ads, More Revs? Is there a Media Bias in the Likelihood to be Reviewed?, June 2012.
- 56 Böckers, Veit, Heimeshoff, Ulrich and Müller Andrea, Pull-Forward Effects in the German Car Scrappage Scheme: A Time Series Approach, June 2012.
- 55 Kellner, Christian and Riener, Gerhard, The Effect of Ambiguity Aversion on Reward Scheme Choice, June 2012.
- 54 De Silva, Dakshina G., Kosmopoulou, Georgia, Pagel, Beatrice and Peeters, Ronald, The Impact of Timing on Bidding Behavior in Procurement Auctions of Contracts with Private Costs, June 2012. Published in: Review of Industrial Organization, 41 (2013), pp.321-343.
- 53 Benndorf, Volker and Rau, Holger A., Competition in the Workplace: An Experimental Investigation, May 2012.
- 52 Haucap, Justus and Klein, Gordon J., How Regulation Affects Network and Service Quality in Related Markets, May 2012. Published in: Economics Letters, 117 (2012), pp. 521-524.
- 51 Dewenter, Ralf and Heimeshoff, Ulrich, Less Pain at the Pump? The Effects of Regulatory Interventions in Retail Gasoline Markets, May 2012.
- 50 Böckers, Veit and Heimeshoff, Ulrich, The Extent of European Power Markets, April 2012.
- 49 Barth, Anne-Kathrin and Heimeshoff, Ulrich, How Large is the Magnitude of Fixed-Mobile Call Substitution? - Empirical Evidence from 16 European Countries, April 2012.
- 48 Herr, Annika and Suppliet, Moritz, Pharmaceutical Prices under Regulation: Tiered Co-payments and Reference Pricing in Germany, April 2012.
- 47 Haucap, Justus and Müller, Hans Christian, The Effects of Gasoline Price Regulations: Experimental Evidence, April 2012.
- Stühmeier, Torben, Roaming and Investments in the Mobile Internet Market, March 2012.
 Published in: Telecommunications Policy, 36 (2012), pp. 595-607.
- 45 Graf, Julia, The Effects of Rebate Contracts on the Health Care System, March 2012, Forthcoming in: The European Journal of Health Economics.
- Pagel, Beatrice and Wey, Christian, Unionization Structures in International Oligopoly, February 2012.
 Published in: Labour: Review of Labour Economics and Industrial Relations, 27 (2013), pp. 1-17.
- Gu, Yiquan and Wenzel, Tobias, Price-Dependent Demand in Spatial Models, January 2012.
 Published in: B. E. Journal of Economic Analysis & Policy, 12 (2012), Article 6.

- 42 Barth, Anne-Kathrin and Heimeshoff, Ulrich, Does the Growth of Mobile Markets Cause the Demise of Fixed Networks? – Evidence from the European Union, January 2012.
- Stühmeier, Torben and Wenzel, Tobias, Regulating Advertising in the Presence of Public Service Broadcasting, January 2012.
 Published in: Review of Network Economics, 11/2 (2012), Article 1.
- 40 Müller, Hans Christian, Forecast Errors in Undisclosed Management Sales Forecasts: The Disappearance of the Overoptimism Bias, December 2011.
- Gu, Yiquan and Wenzel, Tobias, Transparency, Entry, and Productivity, November 2011.
 Published in: Economics Letters, 115 (2012), pp. 7-10.
- Christin, Clémence, Entry Deterrence Through Cooperative R&D Over-Investment, November 2011.
 Published in: Louvain Economic Review, 79/2 (2013), pp. 5-26.
- Haucap, Justus, Herr, Annika and Frank, Björn, In Vino Veritas: Theory and Evidence on Social Drinking, November 2011.
 The theoretical part of this paper is forthcoming as: Haucap/Herr, "A Note on Social Drinking: In Vino Veritas", European Journal of Law and Economics, and the empirical part is forthcoming as: Frank/Haucap/Herr, "Social Drinking Versus Administering Alcohol", Economic Inquiry.
- 36 Barth, Anne-Kathrin and Graf, Julia, Irrationality Rings! Experimental Evidence on Mobile Tariff Choices, November 2011.
- 35 Jeitschko, Thomas D. and Normann, Hans-Theo, Signaling in Deterministic and Stochastic Settings, November 2011. Published in: Journal of Economic Behavior and Organization, 82 (2012), pp.39-55.
- 34 Christin, Cémence, Nicolai, Jean-Philippe and Pouyet, Jerome, The Role of Abatement Technologies for Allocating Free Allowances, October 2011.
- 33 Keser, Claudia, Suleymanova, Irina and Wey, Christian, Technology Adoption in Markets with Network Effects: Theory and Experimental Evidence, October 2011. Published in: Information Economics and Policy, 24 (2012), pp. 262-276.
- 32 Çatik, A. Nazif and Karaçuka, Mehmet, The Bank Lending Channel in Turkey: Has it Changed after the Low Inflation Regime?, September 2011. Published in: Applied Economics Letters, 19 (2012), pp. 1237-1242.
- 31 Hauck, Achim, Neyer, Ulrike and Vieten, Thomas, Reestablishing Stability and Avoiding a Credit Crunch: Comparing Different Bad Bank Schemes, August 2011.
- 30 Suleymanova, Irina and Wey, Christian, Bertrand Competition in Markets with Network Effects and Switching Costs, August 2011. Published in: B. E. Journal of Economic Analysis & Policy, 11 (2011), Article 56.
- Stühmeier, Torben, Access Regulation with Asymmetric Termination Costs, July 2011.
 Published in: Journal of Regulatory Economics, 43 (2013), pp. 60-89.
- 28 Dewenter, Ralf, Haucap, Justus and Wenzel, Tobias, On File Sharing with Indirect Network Effects Between Concert Ticket Sales and Music Recordings, July 2011. Published in: Journal of Media Economics, 25 (2012), pp. 168-178.

- 27 Von Schlippenbach, Vanessa and Wey, Christian, One-Stop Shopping Behavior, Buyer Power, and Upstream Merger Incentives, June 2011.
- 26 Balsmeier, Benjamin, Buchwald, Achim and Peters, Heiko, Outside Board Memberships of CEOs: Expertise or Entrenchment?, June 2011.
- Clougherty, Joseph A. and Duso, Tomaso, Using Rival Effects to Identify Synergies and Improve Merger Typologies, June 2011.
 Published in: Strategic Organization, 9 (2011), pp. 310-335.
- Heinz, Matthias, Juranek, Steffen and Rau, Holger A., Do Women Behave More Reciprocally than Men? Gender Differences in Real Effort Dictator Games, June 2011.
 Published in: Journal of Economic Behavior and Organization, 83 (2012), pp. 105-110.
- 23 Sapi, Geza and Suleymanova, Irina, Technology Licensing by Advertising Supported Media Platforms: An Application to Internet Search Engines, June 2011. Published in: B. E. Journal of Economic Analysis & Policy, 11 (2011), Article 37.
- Buccirossi, Paolo, Ciari, Lorenzo, Duso, Tomaso, Spagnolo Giancarlo and Vitale, Cristiana, Competition Policy and Productivity Growth: An Empirical Assessment, May 2011.
 Published in: The Review of Economics and Statistics, 95 (2013), pp. 1324-1336.
- 21 Karaçuka, Mehmet and Çatik, A. Nazif, A Spatial Approach to Measure Productivity Spillovers of Foreign Affiliated Firms in Turkish Manufacturing Industries, May 2011. Published in: The Journal of Developing Areas, 46 (2012), pp. 65-83.
- 20 Çatik, A. Nazif and Karaçuka, Mehmet, A Comparative Analysis of Alternative Univariate Time Series Models in Forecasting Turkish Inflation, May 2011. Published in: Journal of Business Economics and Management, 13 (2012), pp. 275-293.
- 19 Normann, Hans-Theo and Wallace, Brian, The Impact of the Termination Rule on Cooperation in a Prisoner's Dilemma Experiment, May 2011. Published in: International Journal of Game Theory, 41 (2012), pp. 707-718.
- Baake, Pio and von Schlippenbach, Vanessa, Distortions in Vertical Relations, April 2011.
 Published in: Journal of Economics, 103 (2011), pp. 149-169.
- Haucap, Justus and Schwalbe, Ulrich, Economic Principles of State Aid Control, April 2011.
 Forthcoming in: F. Montag & F. J. Säcker (eds.), European State Aid Law: Article by Article Commentary, Beck: München 2012.
- Haucap, Justus and Heimeshoff, Ulrich, Consumer Behavior towards On-net/Off-net Price Differentiation, January 2011.
 Published in: Telecommunication Policy, 35 (2011), pp. 325-332.
- Duso, Tomaso, Gugler, Klaus and Yurtoglu, Burcin B., How Effective is European Merger Control? January 2011.
 Published in: European Economic Review, 55 (2011), pp. 980-1006.
- Haigner, Stefan D., Jenewein, Stefan, Müller, Hans Christian and Wakolbinger, Florian, The First shall be Last: Serial Position Effects in the Case Contestants evaluate Each Other, December 2010.
 Published in: Economics Bulletin, 30 (2010), pp. 3170-3176.

- Suleymanova, Irina and Wey, Christian, On the Role of Consumer Expectations in Markets with Network Effects, November 2010.
 Published in: Journal of Economics, 105 (2012), pp. 101-127.
- Haucap, Justus, Heimeshoff, Ulrich and Karaçuka, Mehmet, Competition in the Turkish Mobile Telecommunications Market: Price Elasticities and Network Substitution, November 2010.
 Published in: Telecommunications Policy, 35 (2011), pp. 202-210.
- 11 Dewenter, Ralf, Haucap, Justus and Wenzel, Tobias, Semi-Collusion in Media Markets, November 2010. Published in: International Review of Law and Economics, 31 (2011), pp. 92-98.
- 10 Dewenter, Ralf and Kruse, Jörn, Calling Party Pays or Receiving Party Pays? The Diffusion of Mobile Telephony with Endogenous Regulation, October 2010. Published in: Information Economics and Policy, 23 (2011), pp. 107-117.
- 09 Hauck, Achim and Neyer, Ulrike, The Euro Area Interbank Market and the Liquidity Management of the Eurosystem in the Financial Crisis, September 2010.
- Haucap, Justus, Heimeshoff, Ulrich and Schultz, Luis Manuel, Legal and Illegal Cartels in Germany between 1958 and 2004, September 2010.
 Published in: H. J. Ramser & M. Stadler (eds.), Marktmacht. Wirtschaftswissenschaftliches Seminar Ottobeuren, Volume 39, Mohr Siebeck: Tübingen 2010, pp. 71-94.
- 07 Herr, Annika, Quality and Welfare in a Mixed Duopoly with Regulated Prices: The Case of a Public and a Private Hospital, September 2010. Published in: German Economic Review, 12 (2011), pp. 422-437.
- 06 Blanco, Mariana, Engelmann, Dirk and Normann, Hans-Theo, A Within-Subject Analysis of Other-Regarding Preferences, September 2010. Published in: Games and Economic Behavior, 72 (2011), pp. 321-338.
- Normann, Hans-Theo, Vertical Mergers, Foreclosure and Raising Rivals' Costs –
 Experimental Evidence, September 2010.
 Published in: The Journal of Industrial Economics, 59 (2011), pp. 506-527.
- 04 Gu, Yiquan and Wenzel, Tobias, Transparency, Price-Dependent Demand and Product Variety, September 2010. Published in: Economics Letters, 110 (2011), pp. 216-219.
- 03 Wenzel, Tobias, Deregulation of Shopping Hours: The Impact on Independent Retailers and Chain Stores, September 2010. Published in: Scandinavian Journal of Economics, 113 (2011), pp. 145-166.
- 02 Stühmeier, Torben and Wenzel, Tobias, Getting Beer During Commercials: Adverse Effects of Ad-Avoidance, September 2010. Published in: Information Economics and Policy, 23 (2011), pp. 98-106.
- Inderst, Roman and Wey, Christian, Countervailing Power and Dynamic Efficiency, September 2010.
 Published in: Journal of the European Economic Association, 9 (2011), pp. 702-720.

Heinrich-Heine-University of Düsseldorf

Düsseldorf Institute for Competition Economics (DICE)

Universitätsstraße 1_40225 Düsseldorf www.dice.hhu.de